

# AN ANALYSIS OF THE RELATIONSHIP BETWEEN WAGES IN THE PUBLIC AND PRIVATE SECTOR IN COLOMBIA: A PANEL DATA APPROACH

Jesús Otero Luis Fernando Gamboa Andrés García-Suaza

## SERIE DOCUMENTOS DE TRABAJO

No. 97

Abril 2011

## An analysis of the relationship between wages in the public and private sector in Colombia: A panel data approach<sup>\*</sup>

Jesús Otero<sup>†</sup> Facultad de Economía Universidad del Rosario Bogotá, Colombia Luis Fernando Gamboa<sup>‡</sup> Facultad de Economía Universidad del Rosario Bogotá, Colombia

Andrés García-Suaza<sup>§</sup> Facultad de Economía Universidad del Rosario Bogotá, Colombia

April 2011

#### Abstract

This document examines the time-series properties of the wage differentials that arise between the public and private sector in Colombia during the sample period 1984 to 2005. We find conflicting results in unit-root and stationarity tests when looking at wage differentials at an aggregate level (such as for men, women or both). However, when we analyse wage differentials at higher levels of disaggregation, treat them jointly as a panel of data, and allow for the presence of potential cross section dependence, there is more supportive evidence for the view that wage differentials are stationary. This implies that although wage differentials do exist, they have not been consistently increasing (or decreasing) over time.

<sup>\*</sup>The authors gratefully acknowledge the financial support from Centro de Gestión del Conocimiento y la Innovación – Universidad del Rosario (Project FIUR – DVG 080). We would also like to thank Monica Giulietti for useful discussions on several of the ideas developed in the paper. Oscar Avila provided excellent research assistance. Any remaining errors are ours.

<sup>&</sup>lt;sup>†</sup>E-mail: jesus.otero@urosario.edu.co

<sup>&</sup>lt;sup>‡</sup>E-mail: luis.gamboa@urosario.edu.co

<sup>&</sup>lt;sup>§</sup>E-mail: andres.garcia@urosario.edu.co

## 1 Introduction

Historically, the role and the size of the public sector have been topics of considerable debate among economists. In this debate, both employment and remuneration in public sector labour markets have usually been at the core of the discussion. Clearly, these two issues are of importance because of their economic and social policy dimensions, as wages constitute the main source of income for a large number of individuals and their families. In addition, wages serve as the mechanism that guides not only an individual's work-leisure decision problem, but also his/her choice of where to work.

It is well known that economic theory establishes that wages are positively related to labour productivity. However, when it comes to the public sector the question of how wages and employment are determined is not easy to answer. As indicated by Gregory and Borland (1999), while in the private sector employment and earnings are determined within a market environment, in the public sector the same decision-making process takes place within a political environment, where politicians and bureaucrats may well have objective functions that differ from those utilise by the owners of firms in the private sector. Thus, for example, a politician's objective function may well be one of vote-maximisation, or alternatively one of budget-maximisation. In addition to this, the effects of unions and collective bargaining are more likely to be observed in public sector labour markets than in private ones.

The purpose of this document is to study the relationship between wages in the public and private sector in Colombia. During the last two decades or so, Colombia has become an interesting case-study that features high and persistent unemployment, low rates of job creation in the formal private sector, and the growing importance of an informal private sector that is characterised by new contractual forms in the labour market.<sup>1</sup> These new contractual forms include the creation of

<sup>&</sup>lt;sup>1</sup>Lora and Márquez (1998) present some stylised facts of employment in Colombia as well as in other Latin American economies see . For a formal econometric analysis of the labour market in Colombia see e.g. Arango and Posada (2002).

companies whose main objective is to act as intermediaries between workers and firms, where the latter aim to minimise the payroll taxes associated to the formal sector, which include social security payments and pension contributions, among others. As a result, jobs in the public sector become more attractive due to the fact that it is more difficult for employers to avoid the payment of these additional costs to their employees.

International literature on the evolution of wages in the public and private sectors and the corresponding differential that arises between them is extensive; see, inter alia, Pederson, Schmidt-Sørensen, Smith, and Westergård-Nielsen (1990), Hundley (1991), Mueller (1998), Tansel (1998), Adamchik and Bedi (2000), Bender (2002), Panizza and Qiang (2005) and Lamo, Perez, and Schuknecht (2008). Literature for Colombia is more scarce. Two relatively recent exceptions are Arango and Posada (2007), who present a descriptive analysis of the dynamic behaviour of public sector wages in several occupational categories, and Galvis (2010), who studies wage disparities for private sector employees. Based on our literature review, it appears that there are no other works for Colombia that analyse the behaviour of public and private sector wages, both within each sector as well as between them.

The main objective of this document is to examine whether there is evidence of a stable long-run equilibrium relationship between wages in the public and private sector in Colombia. We believe that the study of the Colombian experience is interesting because ever since 1886, the central government concentrated political, administrative and fiscal powers. Then, in 1991 a constitutional reform introduced important modifications and changes in the existing territorial order of the country, so that regional and local levels of government were given greater power and responsibilities, and a new set of parameters to assign and determine transfers from the central government were defined.<sup>2</sup> Thus, it is of some interest to determine whether this decentralisation process has led to increasing wage differentials between sectors. Initially, this analysis can be undertaken at an aggregate level by comparing

 $<sup>^{2}</sup>$ See e.g. Iregui (2005), for estimates of the welfare effects of decentralisation in Colombia.

earnings in the private sector with those in the public sector. Then, one could further analyse gender, city and occupational category differentials between public and private sector employees.

This document offers three main contributions to existing literature. First, we use data from Colombian household surveys collected over the period 1984 to 2005. The advantage of this source of information is that it allows us to focus on how individual-level data have changed over time for different economic sectors, gender, occupational category and city. The selection of the sample period is dictated by the availability of a consistent dataset. This is because in 2006 significant methodological changes were implemented in the household survey system, so that results before and after this year are not directly comparable.

Second, we focus on an examination of the time series properties of wage differentials or, put another way, we assess whether or not wage differentials are stationary. In this sense, finding that a wage differential is stationary is equivalent to saying that the two wages are cointegrated with a known cointegrating vector equal to [1, -1]'. From an economic point of view, this means that wage levels maintain a stable long-run equilibrium relationship, so that the corresponding wage differential does not increase (or decrease) without bound as time passes. To the best of our knowledge, this empirical modelling approach has not been implemented by other authors.

Third, the time-series analysis will be undertaken by looking at each wage differential individually as well as within a panel data framework. The advantage of adopting a panel data setup is that it allows us to examine the potential effect of cross-sectional dependence among wages that may arise from common shocks (or innovations). Another advantage of panel data is that by combining information from the time-series dimension with that from the cross-section dimension, fewer time series observations are required for statistical tests to have power.

The document is organised as follows. Section 2 presents a brief literature review on wage differentials between the public and private sectors. Section 3 presents the methodology that will be used in the document. We start off by briefly presenting important time-series concepts that will be used in the document, along with their economic implications. Then, we describe the statistical tests that will be applied to assess the time-series properties of several wage differentials. Section 4 describes the data that will be used in the document and shows some of the stylised facts about the evolution of wages in Colombia. Section 5 examines the time-series properties of wage differentials in the country. Section 6 offers concluding remarks.

## 2 Brief literature review

During the last three decades or so, a number of authors have analysed the dynamic behaviour of public and private sector wages. The study of wage differentials between the public and private sectors has been motivated, among other factors, by the recent growth of the public sector in many countries around the world and the corresponding cost-implications on tax-payers. There are two main reasons why one should be interested in the operation of public sector labour markets. First, public sector labour markets are large and their financial resources primarily come from the functioning of the private sector. Second, public sector labour markets are different from private sector labour markets in as much as politicians or bureaucrats' objective function differs from that of the owners of private sector firms. Indeed, as indicated in the previous section, decision-making on public sector employment and wages takes place in a political environment, where politicians and bureaucrats may have objectives that not always seek profitability. By contrast, private sector decision-making occurs in a market environment in which owners (or shareholders) of private sector firms continuously monitor the performance of their companies.

Ehrenberg and Schwarz (1986), Gregory and Borland (1999) and Bender (2002) present comprehensive surveys of the literature concerning wage differentials between the public and private sectors. Ehrenberg and Schwarz (1986) focus mainly on studies dealing with the United States, while the other two also include references for other developed and developing countries. Bender (2002) reviews the literature that provides theoretical reasons that attempt to explain the existence of public/private wage differentials. According to this author, there are several factors that may be taken into consideration when examining possible explanations. First, there is the influence of trade unions on demand for public sector goods and the 'vote producing' activities by civil servants. Second, part of the public/private wage differential may emerge from economic rents perceived by public sector workers because of their bargaining power in those services that are considered essential. Third, the idea of a public/private wage differential has to be analysed carefully because of the existence of selection bias and other econometric problems that may arise from the available data. According to Bender (2002) a common finding of the empirical studies in his survey is the existence of a declining premium paid to central government employees, although for developing countries wage differentials are found to be negative in some instances. As will be shown below, there are different ways to study the existence of wage differentials between the public and private sectors. However, an important factor that must be taken into account, in particular for the purposes of international comparisons, is the size of the public sector in the economy, as it reflects the capacity of the sector to compete for workers in the labour market.

In their analysis of public sector labour markets, Gregory and Borland (1999) find a persistent increase in the size of the public sector in several countries. According to these authors, public sector workers get higher average earnings than private sector ones due to differences in levels of education (which is higher for individuals in the public sector). At the same time, these authors also find that the earnings distribution of public sector workers exhibits is more concentrated around its mean value compared to that of private sector workers. In addition, the union/non-union and male/female wage gaps tend to be smaller in the public sector.

Turning to specific country-case studies, most of the literature is related to developed and emerging market countries; see, for instance, Pederson, Schmidt-Sørensen, Smith, and Westergård-Nielsen (1990) for Denmark, Hundley (1991) for the United States, Alvarez, Jareño, and Sebastian (1993) for Spain, Dustmann and Soest (1998) for Germany, Mueller (1998) for Canada, Tansel (1998) for Turkey, Adamchik and Bedi (2000) for Poland, and Lamo, Perez, and Schuknecht (2008) for a sample of OECD and Euro zone countries. Some of these studies have looked at the determinants of wages, while others have examined the existence of wage differentials between the public and private sectors.

Pederson, Schmidt-Sørensen, Smith, and Westergård-Nielsen (1990) examine the public/private wage differential using Danish data from a panel of individuals over the period 1976 to 1985. The results of estimating fixed-effect type regressions show evidence that a wage-twist policy has been applied over the sample period. The idea of a wage-twist policy is to implement a series of mechanisms to affect the allocation of resources between the public and the private sector. Thus, for example, a government may attempt to reduce the private/public wage differential in order to overcome recruitment and retention problems for public sector employees. An additional interesting finding form their work is that "women employed in the formal sector tend to have higher average skills than men employed in the formal sector (...) probably due to supply and demand considerations" (p. 830). According to these authors, if it is accepted that job security is important in the public sector, then the public wage premium has to be analysed from a different perspective, such as the general equilibrium one considered by Shapiro and Stiglitz (1984).

Hundley (1991), using data from the 1985 Current Population Survey of the United States, finds that public/private wage differentials tend to decline as the level of skill required to fulfil an occupational category is increased. Mueller (1998) uses quantile regression techniques to estimate the size of the public/private wage differential in Canada. This author finds that this differential tends to be highest for women, federal government employees, and individuals at the lower tail of the wage distribution. The use of quantile regressions is crucial to understand differences in public/private wages over the whole distribution of wages and not only at the mean of the distribution.

Dustmann and Soest (1998) analyse several statistical assumptions used in em-

pirical models on public/private sector wage structures for Germany. They use an extended version of a standard switching regression model that allows for endogeneity in the level of education, experience, and hours worked. Several model specifications are estimated and the results of such models are subsequently compared. It turns out that their results are sensitive to the identification assumptions that are adopted, but robust to the regressors that are included in the model. Alvarez, Jareño, and Sebastian (1993) use Spanish data over the period 1964–1991 to analyse transmission mechanisms between prices and nominal wages. The results of estimating a VAR model indicates that private sector wages help explain inflation, while public sector wages play a minor role. They also find that public sector wages do not have an impact on private sector wages.

Adamchik and Bedi (2000) examine whether there are wage differentials between workers in the public and the private sectors in Poland. After standardising for worker characteristics and sector selection effects, they find evidence of a private sector positive wage premium. This wage premium is particularly large for university educated workers. According to the authors, the existence of these wage differentials make it difficult for the public sector to attract and retain skilled employees. In addition, lower public sector wages may encourage moonlighting and compromise the efficiency of the public sector.

Lamo, Perez, and Schuknecht (2008) analyse the interaction between public and private sector wages during 1960-2006 for a sample of OECD and Euro zone countries. They find empirical evidence that salaries are positively correlated over the business cycle. Also, they uncover evidence of short-, medium- and long-run comovements between public and private sector salaries. Lastly, causality tests suggest a predominant role of private salaries over the business cycle. Tansel (1998) finds mixed results when analysis public/private wage differential in Turkey. Indeed, after controlling for observed individual characteristics and sample selection bias, this author finds that wages for men (women) in public administration are lower (greater) than wages for men (women) in the private sector. Moreover, wage differentials because of gender are not found for individuals working in the public sector, although they are found to exist for workers in the private sector.

For Latin American countries, existing literature appears more scarce; see e.g. Panizza and Qiang (2005) for a study of a sample of Latin American countries, Stelcner, der Gaag, and Vijverberg (1989) for the case of Peru, and Arango and Posada (2007) and Galvis (2010) for studies of the Colombian case. Panizza and Qiang (2005) use household survey data for thirteen Latin American countries to investigate wage differentials between the public and private sectors, as well as wage differentials that may arise because of gender.<sup>3</sup> These authors find that in the cases of Brazil, Colombia, Costa Rica, Ecuador and El Salvador, there is evidence of a statistically and economically significant public wage premium that favours male workers. Interestingly, in the cases of Brazil, Colombia, Costa Rica, Honduras, Mexico and El Salvador, they also uncover evidence of a wage premium for women working in the public sector. Stelcner, der Gaag, and Vijverberg (1989) use Peruvian data to study the emergence of wage differentials between male and female workers in the public and private sectors. The authors estimate a switching regression model and find that there is not a "pure" wage advantage or economic rent of government workers when corrected estimates of wage functions are compared. Their findings indicate that in Lima there is a wage differential in favour of private sector employees, while in other urban areas there is no significant wage differential. In the case of Colombia, the evolution of the public and private sector wages has not received a great deal of attention. Two relatively recent exceptions are Arango and Posada (2007) and Galvis (2010). Arango and Posada (2007) present a descriptive analysis of the dynamic behaviour of public sector wages in several occupational categories. An interesting feature of this work is that the authors use payroll information from the Ministry of Finance, which covers the sample period 1978–2005. On the other hand, Galvis (2010) uses Colombian household survey data for the period 1984 to 2009, to study real wage disparities for private sector employees in the seven main

<sup>&</sup>lt;sup>3</sup>These countries are Bolivia, Brazil, Colombia (urban only), Costa Rica, El Salvador, Ecuador, Honduras, Mexico, Nicaragua, Panama, Paraguay, Peru and Venezuela.

metropolitan areas, namely Barranquilla, Bogotá, Bucaramanga, Cali, Manizales, Medellín and Pasto. This author finds that there are significant differentials among different categories of private sector wages, and that in some cases these differentials have been growing over time.

## 3 Methodology

#### **3.1** Background

Economists typically use information that is available in three different formats: (i.) cross-section data; (ii.) time-series data; and (iii.) panel data. Cross-section data describe the activities of individuals, firms, countries or other units of analysis that are collected at a particular moment in time. Time-series data describe the movement of a variable through time, and this could be for different periodicity such as annual, quarterly, monthly, weekly or even daily. Lastly, panel data combine the first two types of information; that is, they describe the activities of individuals, firms, countries or other units of analysis through time.

Focusing for the moment in the analysis of time-series data, the starting point is the concept of stationarity. A time-series is said to be stationary if its probability distribution function does not change through time. In practice, this definition turns out to be very strong and difficult to verify, so that a weaker version of the concept states that a time-series is "weakly stationary" (also referred to as "covariance stationary") if its first two moments (i.e. its mean and its variance) do not change over time. The intuition behind this definition is that if the first two moments of a time-series do not change through time, then its future is going to be similar to its past, and this can be exploited for forecasting purposes.

Unfortunately, most to the time-series that are studied by economists and econometricians alike, exhibit a tendency to increase over time, that is, they are nonstationary series. This is case, for example of the series of the gross domestic product (both in nominal and real terms), consumer price index, monetary aggregates, employment, and wages, among others. From a statistical point of view, the use of non-stationary series involves a number of problems in the sense that conventional hypothesis tests, confidence intervals and predictions about the future are not going to be reliable. A classic example of the problems that arise with the use of non-stationary data is that of the non-sense regression problem discovered by Yule (1926), or spurious regression problem in the terminology of Granger and Newbold (1974). The idea is that when one estimates a regression between two or more variables that grow over time (whatever these variables may be), it is more likely than not that one is going to find an apparent positive association between the variables, regardless of whether this association truly exists.<sup>4</sup>

The solution to the problem of working with non-stationary series depends upon the nature or cause of the non-stationarity, that is whether it is deterministic or stochastic. Until the early 1980s, the trending behaviour of a time-series used to be eliminated by running a regression of the series under consideration against a linear (or even a polynomial) time trend. What this statistical approach does not recognise, however, is that it is only valid when the rate of growth of the series is always the same or, in other words, when it is constant. If, on the other hand, the rate of growth of the series is not always the same (or if it cannot be predicted perfectly), then the time series is said to have a stochastic trend, in which case the suitable approach to eliminate the non-stationarity would be to calculate the first difference of the series.

In time-series terminology, a series that needs to be differenced in order to become stationary is said to be a series integrated of order 1, and is denoted ~ I(1) for short. In general, a series that needs to be differenced d times in order to become stationary is said to be a series integrated of order d, and is denoted ~ I(d). In summary, economic time-series that exhibit a trending behaviour can be classified in two groups. The first group is that of the series that are stationary around a linear

<sup>&</sup>lt;sup>4</sup>Examples of the spurious regression problem include Yule (1926) who, using annual information for the period 1866–1911, finds a correlation coefficient of 95% between mortality rate in Enlgand and Wales and the proportion of Church of England marriages. Another example is provided by Hendry (1980) who finds a correlation coefficient of 98% between inflation and cumulative rainfall in the UK.

or polynomial trend, which are referred to as Trend Stationary Processes (TSP). The second group is that of the series that need to be differenced one or many more times. Interestingly, it should be noticed that a stationary series is integrated of order 0, denoted  $\sim I(0)$ , since it is not necessary to difference the series to make it stationary.

The fact that a time-series can be classified as an  $\sim I(0)$  or an  $\sim I(1)$  process can have important implications from an economic point of view. Indeed, in the case of an  $\sim I(0)$  series, a shock (or innovation) will have a temporary effect (i.e. the effect of the shock disappears as time passes) and out-of-sample predictions are more precise. On the contrary, in the case of an  $\sim I(1)$  series, the effect of a shock (or innovation) is permanent and out-of-sample predictions are less accurate; see e.g. Franses (1998). More importantly, while the variance an  $\sim I(0)$  series is finite, that of an  $\sim I(1)$  series increases without a bound.

In the specific case of public/ private sector wage differentials, which are the purpose of analysis of this document, finding that they can be best characterised as ~ I(0) processes suggests that there would not be a tendency for them to consistently increase (or decrease) over time. On the contrary, if public/ private sector wage differentials are found to be ~ I(1), then there would be arbitrage opportunities for individuals by moving from one sector to the other. This, of course, may become a serious obstacle for governments that aim to attract and maintain a productive labour force in the public sector.

#### **3.2** Unit-root and stationarity tests

During the late 1970s and early 1980s, Dickey and Fuller (1979) and Dickey and Fuller (1981) propose statistical tests to determine whether a time series can be best characterised as TSP or DSP. These statistical tests are referred to in the literature as unit-root tests.

Dickey and Fuller consider three different models to test for a unit root:

$$y_t = \phi_1 y_{t-1} + \epsilon_t, \tag{1}$$

$$y_t = \phi_0 + \phi_1 y_{t-1} + \epsilon_t, \tag{2}$$

$$y_t = \phi_0 + \phi_1 y_{t-1} + \phi_2 t + \epsilon_t, \tag{3}$$

where  $y_t$  is the variable of interest (in our case a wage differential) for which we have t = 1, ..., T available observations. These equations can be estimated by ordinary least squares (OLS), and the null hypothesis to test the presence of a unit root is  $H_0: \phi_1 = 1$ , against the alternative that the series is stationary  $H_a: \phi_1 < 1$ .

For practical purposes, model (1) is rarely estimated as it is too restrictive because it assumes that  $y_t$  has a mean of zero (the model is very important for theoretical purposes though). In turn, model (2) is estimated when  $y_t$  has a nonzero mean, while model (3) is applied whenever  $y_t$  exhibits an upward (or downward) behaviour; for a more formal sequential testing procedure see Perron (1988). The models considered by Dickey and Fuller can be alternatively reparameterised as:

$$\Delta y_t = (\phi_1 - 1) y_{t-1} + \epsilon_t,$$

$$\Delta y_t = \phi_0 + (\phi_1 - 1) \, y_{t-1} + \epsilon_t,$$

$$\Delta y_t = \phi_0 + (\phi_1 - 1) y_{t-1} + \phi_2 t + \epsilon_t,$$

where  $\Delta y_t$  is the first difference of the series of interest, i.e.  $\Delta y_t = y_t - y_{t-1}$ . Notice that testing the null hypothesis that  $H_0: \phi_1 = 1$  is equivalent to test  $H_0: b = 0$  in the following models:

$$\Delta y_t = by_{t-1} + \epsilon_t,\tag{4}$$

$$\Delta y_t = \phi_0 + by_{t-1} + \epsilon_t, \tag{5}$$

$$\Delta y_t = \phi_0 + by_{t-1} + \phi_2 t + \epsilon_t. \tag{6}$$

To perform the unit-root test, one calculates the t-statistic associated to the estimated coefficient on  $y_{t-1}$ , which is then compared with the critical values tabulated by Fuller (1976), Dickey and Fuller (1981), or the more recent response surfaces estimated by MacKinnon (1991).

An important assumption behind the construction of the Dickey and Fuller tests is that of no serial correlation, i.e.  $\epsilon_t \sim iid(0, \sigma^2)$ . If this assumption does not hold, then Said and Dickey (1984) suggest introducing lags of the dependent variable in order to whiten the residuals. Including lags of the dependent variable in equations (4), (5) and (6) yields:

$$\Delta y_t = by_{t-1} + \sum_{i=1}^p \delta_i \Delta y_{t-i} + \epsilon_t,$$
  
$$\Delta y_t = \phi_0 + by_{t-1} + \sum_{i=1}^p \delta_i \Delta y_{t-i} + \epsilon_t,$$
  
$$\Delta y_t = \phi_0 + by_{t-1} + \phi_2 t + \sum_{i=1}^p \delta_i \Delta y_{t-i} + \epsilon_t.$$

These three equations are referred to as the (Augmented) Dickey and Fuller (1979) (ADF) test regressions.

Kwiatkowski, Phillips, Schmidt, and Shin (1992) (KPSS) propose an alternative approach, in which the null and alternative hypotheses are interchanged. That is, they propose a residual-based Lagrange Multiplier (LM) tests for the null hypothesis that a time series is stationary (either around a level or a deterministic time trend), against the alternative that it is non-stationary.

KPSS consider the model:

$$y_t = \beta t + r_t + \phi y_{t-1} + \varepsilon_t,$$

where  $r_t = r_{t-1} + u_t$ ,  $\varepsilon_t$  is  $iid(0, \sigma_{\varepsilon}^2)$ ,  $\sigma_{\varepsilon}^2 = 1$ ,  $u_t$  is  $iid(0, \sigma_u^2)$ , and  $|\phi| < 1$ .

Two cases of interest arise in the previous model setup. The first one occurs when  $\sigma_u^2 = 0$ , and the initial value of  $r_t$  is assumed to be fixed and equal to  $r_0$ . In this case,  $y_t$  is a stationary series around a linear trend term (it should be recalled that  $|\phi| < 1$ ). Notice that if one further assumes that  $\beta = 0$ , then  $y_t$  is a stationary series around a mean. The second one arises when  $\sigma_u^2 > 0$ . In this case,  $y_t$  becomes non-stationary. Thus, the previous discussion implies that the KPSS test of the null hypothesis that a series is stationary is given by  $H_0: \sigma_u^2 = 0$ , while the alternative hypothesis that it is non-stationary can be stated as  $H_a: \sigma_u^2 > 0$ .

The test statistic constructed by KPSS is:

$$\eta_j = \frac{T^{-2} \sum_{t=1}^T S_t^2}{s_T^2(l)}; \qquad j = 1, 2$$

where  $S_t = \sum_{k=1}^t \hat{\varepsilon}_k$  is the partial sum of the residuals  $(\hat{\varepsilon}_k)$  that result from running a regression of  $y_t$  against an intercept, for j = 1, or a regression of  $y_t$  against an intercept and a linear trend term, for j = 2, depending on whether the null hypothesis of interest is that of stationarity around a mean, or around a linear deterministic trend, respectively.

In addition,  $s_T^2(l)$  is an estimator of the long-run variance of the corresponding regression. In their original paper, KPSS propose a non-parametric estimator of  $\hat{\sigma}_{\varepsilon_i}^2$  based on a Bartlett window having a truncation lag parameter of  $l_q =$ integer  $\left[q \left(T/100\right)^{1/4}\right]$ , with q = 4, 12. However, Caner and Kilian (2001) have pointed out that stationarity tests, like the KPSS tests, exhibit very low power after correcting for size distortions. Thus, in our paper we follow recent work by Sul, Phillips, and Choi (2005), who propose a new boundary condition rule that improves the size and power properties of the KPSS stationarity tests. In particular, Sul et al. suggest the following procedure. First, an AR(p) model for the residuals is estimated, that is:

$$\hat{\varepsilon}_t = \rho_1 \hat{\varepsilon}_{t-1} + \dots + \rho_p \hat{\varepsilon}_{t-p} + v_t, \tag{7}$$

where the lag length of the autoregression can be determined for example using the GEneral-To-Specific (GETS) algorithm proposed by Hall (1994) and Campbell and Perron (1991). Second, the long-run variance estimate of  $\hat{\sigma}_{\varepsilon}^2$  is obtained with the boundary condition rule:

$$\hat{\sigma}_{\varepsilon}^{2} = \min\left\{T\hat{\sigma}_{\upsilon}^{2}, \frac{\hat{\sigma}_{\upsilon}^{2}}{\left(1 - \hat{\rho}\left(1\right)\right)^{2}}\right\},\tag{8}$$

where  $\hat{\rho}(1) = \hat{\rho}_1(1) + ... + \hat{\rho}_p(1)$  denotes the autoregressive polynomial evaluated at L = 1. In turn,  $\hat{\sigma}_v^2$  is the long-run variance estimate of the residuals in equation (14) that is obtained using a quadratic spectral window Heteroskedastic and Autocorrelation Consistent (HAC) estimator.<sup>5</sup>

It is worth mentioning that although other unit-root tests are available in the literature, see e.g. Maddala and Kim (1998) for a textbook exposition, in this document we focus on the ADF and KPSS tests since they have already been extended to deal with panel data. In the next section we briefly review the panel unit-root and stationarity tests that will be applied in this document.

#### **3.3** Unit-root and stationarity tests in panel data

The problem of testing the presence of unit roots in panels of data has received a great deal of attention in recent years; see e.g. the literature reviews in Breitung and Pesaran (2008) and Banerjee and Wagner (2009). Among the tests available in the literature, the Im, Pesaran, and Shin (2003) (IPS) test has proved to be the most popular. This panel unit root test combines information from the time-series dimension with that from the cross-section dimension, such that fewer time

<sup>&</sup>lt;sup>5</sup>Additional Monte Carlo evidence reported by Carrión-i Silvestre and Sansó (2006) also indicates that the proposal in Sul, Phillips, and Choi (2005) is to be preferred since the KPSS statistics exhibit less size distortion and reasonable power.

observations are required for the test to have power. The IPS test is based on individual ADF test regressions:

$$\Delta y_{it} = a_i + b_i y_{i,t-1} + \sum_{r=1}^{p_i} c_{ir} \Delta y_{i,t-r} + \varepsilon_{it}, \qquad (9)$$

where  $y_{it}$  denotes relative average wage (per hour) for individual i = 1, ..., N, at time period t = 1, ..., T. In this setting, the null hypothesis to test the presence of a unit root becomes  $H_0: b_i = 0$  for all i, against the alternative that at least one of the individual series in the panel is stationary, that is  $H_1: b_i < 0$  for at least one i. The IPS test averages the ADF statistics obtained in equation (9) across the Ncross-sectional units of the panel, denoted as:

$$t_{bar_{NT}} = (N)^{-1} \sum_{i=1}^{N} t_{i,T},$$

where  $t_{i,T}$  is the ADF test for the  $i^{th}$  cross-sectional unit in the panel. IPS show that after a suitable standardisation, the  $t_{bar_{NT}}$  statistic follows a standard normal distribution. Moreover, they compute the mean and variance required to standardise the  $t_{bar_{NT}}$  statistic via Monte Carlo simulations, for different values of T and  $p_i$ , and for different combinations of deterministic components; that is, when the test regression (9) includes intercept but no trend, and when it includes both intercept and trend.

An important assumption underlying the IPS test is that of cross section independence among the individual time series in the panel. However, it has been shown that the test suffers from size distortions in the presence of cross section dependence, the magnitude of which increases as the strength of the cross-sectional dependence increases; see e.g. Strauss and Yigit (2003) and Pesaran (2007). To allow for the presence of cross section dependence, Pesaran (2007) proposes to augment equation (9) with the cross-sectional averages of lagged level and lagged first-differences of the individual series in the panel. Thus, the test of the unit root hypothesis would be based on the following  $p^{th}$  order Cross-sectionally Augmented ADF (denoted CADF) regressions:

$$\Delta y_{it} = a_i + b_i y_{i,t-1} + \sum_{r=1}^p c_{ir} \Delta y_{it-r} + d_i \bar{y}_{t-1} + \sum_{r=0}^p f_{ir} \Delta \bar{y}_{t-r} + \varepsilon_{it}, \qquad (10)$$

where  $\bar{y}_t$  is the cross section mean of  $y_{it}$ , defined as  $\bar{y}_t = (N)^{-1} \sum_{i=1}^{N} y_{it}$ . The corresponding cross-sectionally augmented version of the IPS test statistic (denoted CIPS) is given by:

$$CIPS = (N)^{-1} \sum_{i=1}^{N} \tilde{t}_i,$$

where  $\tilde{t}_i$  is the cross-sectional ADF statistic for the  $i^{th}$  individual in the panel. Once again, under the null hypothesis there is a unit root in all individuals in the panel, i.e.  $H_0: b_i = 0$  for all i, while under the alternative at least one of the individual series in the panel is stationary, i.e.  $H_1: b_i < 0$  for at least one i. The critical values of the CIPS statistic are tabulated via Monte Carlo simulations by Pesaran (2007) for various values of T and N and according to the deterministic elements included in the cross-sectionally augmented ADF regressions, namely no intercepts and no trends (Case I), intercepts only (Case II), and intercepts and trends (Case III).

An important issue that arises when using both the IPS and CIPS tests is that due to the heterogeneous nature of the alternative hypothesis, one needs to be careful when interpreting the results, because the null hypothesis that there is a unit root in each cross section may be rejected when only a fraction of the series in the panel are stationary. By contrast, Hadri (2000) proposes residual-based LM tests for the null hypothesis that all the time series in the panel are stationary (either around a level or a deterministic time trend), against the alternative that some of the series are non-stationary. The Hadri tests thus offer the advantage that if the null hypothesis is not rejected, there would be evidence that all wage differentials in the panel are stationary.

Following Hadri (2000), consider the models:

$$y_{it} = r_{it} + \varepsilon_{it} \tag{11}$$

and

$$y_{it} = r_{it} + \beta_i t + \varepsilon_{it} \tag{12}$$

where  $r_{it}$  is a random walk,  $r_{it} = r_{i,t-1} + u_{it}$ , and  $\varepsilon_{it}$  and  $u_{it}$  are mutually independent normal distributions. Also,  $\varepsilon_{it}$  and  $u_{it}$  are *i.i.d* across *i* and over *t*, with  $E[\varepsilon_{it}] = 0$ ,  $E[\varepsilon_{it}^2] = \sigma_{\varepsilon,i}^2 > 0$ ,  $E[u_{it}] = 0$ ,  $E[u_{it}^2] = \sigma_{u,i}^2 \ge 0$ , t = 1, ..., T and i = 1, ..., N. The null hypothesis that all the series are stationary is given by  $H_0: \sigma_{u,i}^2 = 0$ , i = 1, ..., N, while the alternative that some of the series are non-stationary is  $H_1: \sigma_{u,i}^2 > 0$ ,  $i = 1, ..., N_1$  and  $\sigma_{u,i}^2 = 0$ ,  $i = N_1 + 1, ..., N$ .

Let  $\hat{\varepsilon}_{it}$  be the residuals from the regression of  $y_{it}$  on an intercept, for model (11) (or on an intercept and a linear trend term, for model (12)). Then, for individual *i* the univariate KPSS stationarity test is:

$$\eta_{i,T} = \frac{\sum_{t=1}^{T} S_{it}^2}{T^2 \hat{\sigma}_{\varepsilon_i}^2},\tag{13}$$

where  $S_{it}$  denotes the partial sum process of the residuals given by  $S_{it} = \sum_{j=1}^{t} \hat{\varepsilon}_{ij}$ , and  $\hat{\sigma}_{\varepsilon_i}^2$  is a consistent estimator of the long-run variance of  $\hat{\varepsilon}_{it}$  from the appropriate regression, for which we follow the procedure suggested by Sul, Phillips, and Choi (2005). This procedure was outlined earlier in a univariate context. Within a panel framework, the procedure advocated by Sul et al. is implemented as follows: First, for each individual *i* an AR model for the residuals is estimated, that is:

$$\hat{\varepsilon}_{it} = \rho_{i,1}\hat{\varepsilon}_{i,t-1} + \dots + \rho_{i,p_i}\hat{\varepsilon}_{i,t-p_i} + \upsilon_{it},\tag{14}$$

where the lag length of the autoregression can be determined for example using the GETS algorithm proposed by Hall (1994) and Campbell and Perron (1991). Second, the long-run variance estimate of  $\hat{\sigma}_{\varepsilon_i}^2$  is obtained with the boundary condition rule:

$$\hat{\sigma}_{\varepsilon_i}^2 = \min\left\{T\hat{\sigma}_{\upsilon_i}^2, \frac{\hat{\sigma}_{\upsilon_i}^2}{\left(1 - \hat{\rho}_i\left(1\right)\right)^2}\right\},\tag{15}$$

where  $\hat{\rho}_i(1) = \hat{\rho}_{i,1}(1) + ... + \hat{\rho}_{i,p_i}(1)$  denotes the autoregressive polynomial evaluated at L = 1. In turn,  $\hat{\sigma}_{v_i}^2$  is the long-run variance estimate of the residuals in equation (14) that is obtained using a quadratic spectral window HAC estimator.<sup>6</sup>

The Hadri (2000) panel stationarity test statistic is given by the simple average of individual univariate KPSS stationarity tests:

$$\widehat{LM}_{T,N} = \frac{1}{N} \sum_{i=1}^{N} \eta_{i,T},$$

which after a suitable standardisation, using appropriate moments, follows a standard normal limiting distribution. That is:

$$Z = \frac{\sqrt{N}\left(\widehat{LM}_{T,N} - \overline{\xi}\right)}{\overline{\zeta}} \Rightarrow N(0,1), \qquad (16)$$

where  $\bar{\xi} = \frac{1}{N} \sum_{i=1}^{N} \xi_i$  and  $\bar{\zeta}^2 = \frac{1}{N} \sum_{i=1}^{N} \zeta_i^2$  are respectively the mean and variance required for standardisation. Asymptotic values of these moments can be found in Hadri (2000), while finite sample critical values appear in Hadri and Larsson (2005).

The Monte Carlo experiments of Hadri (2000) illustrate that these tests have good size properties for T and N sufficiently large. However, Giulietti, Otero, and Smith (2009) show that even for relatively large T and N the Hadri (2000) tests suffer from severe size distortions in the presence of cross-sectional dependence, the magnitude of which increases as the strength of the cross-sectional dependence increases. To correct for the size distortion caused by cross-sectional dependence, Giulietti et al. apply the bootstrap method and find that the bootstrap Hadri tests are approximately correctly sized.

To implement the bootstrap method in the context of the Hadri tests, we start off by correcting for serial correlation using equation (14) and obtain  $\hat{v}_{it}$ , which are centred around zero. Next, as in Maddala and Wu (1999), the residuals  $\hat{v}_{it}$  are resampled with replacement with the cross-section index fixed, so that their crosscorrelation structure is preserved; the resulting bootstrap innovation  $\hat{v}_{it}$  is denoted  $\hat{v}_{it}^*$ . Then,  $\hat{\varepsilon}_{it}^*$  is generated recursively as:

<sup>&</sup>lt;sup>6</sup>Additional Monte Carlo evidence reported by Carrión-i Silvestre and Sansó (2006) also indicates that the proposal in Sul, Phillips, and Choi (2005) is to be preferred since the KPSS statistics exhibit less size distortion and reasonable power.

$$\hat{\varepsilon}_{it}^* = \hat{\rho}_{i,1}\hat{\varepsilon}_{i,t-1}^* + \ldots + \hat{\rho}_{i,p_i}\hat{\varepsilon}_{i,t-p_i}^* + \upsilon_{it}^*,$$

where, in order to ensure that initialisation of  $\hat{\varepsilon}_{it}^*$ , i.e. the bootstrap samples of  $\hat{\varepsilon}_{it}$ , becomes unimportant, we follow Chang (2004) who advocates generating a large number of  $\hat{\varepsilon}_{it}^*$ , say T + Q values and discard the first Q values of  $\hat{\varepsilon}_{it}^*$  (for our purposes we choose Q = 40). Lastly, the bootstrap samples of  $y_{it}^*$  are calculated by adding  $\hat{\varepsilon}_{it}^*$  to the deterministic component of the corresponding model, and the Hadri LM statistic is calculated for each  $y_{it}^*$ . The previous steps are repeated several times in order to derive the empirical distribution of the LM statistic, from which bootstrap probability values (or alternatively bootstrap critical values) may be obtained.

### 4 Data

We use data from the nationwide household surveys periodically undertaken by the Departamento Administrativo Nacional de Estadística (DANE). Our period of analysis, which runs from 1984 to 2005, is characterised by the implementation of two different surveys, namely the Encuesta Nacional de Hogares (ENH) and the Encuesta Continua de Hogares (ECH). The former was applied quarterly from 1979 to 2000, and up to 1983 included the four main cities: Bogotá, Medellín, Cali and Barranquilla. In 1984 three more cities were added to the ENH: Bucaramanga, Manizales and Pasto. In 2001, the ENH was superseded by the ECH, which is a monthly survey of thirteen cities: the original seven plus Ibagué, Montería, Cartagena, Pereira, Villavicencio and Cúcuta.<sup>7</sup>

The dataset used in the analysis consists of the hourly wage per worker. The data for each year in the period 1984-2005 was obtained by aggregating the surveys of that year. Appendix 1 reports the number of observations used in the analysis. Appendix 2 presents the series of hourly wage per worker in current pesos. These

<sup>&</sup>lt;sup>7</sup>The ECH also introduced changes in the phrasing of questions aimed at measuring labour market indicators, such us the concept of unemployment, unpaid workers, etc. These methodological differences do not affect our measure of hourly average wage.

data are subsequently deflated by the overall consumer price index (2005=100) to account for the effect of inflation; see Appendix 3.

The analysis of the evolution of wage differentials can be performed from different perspectives. To begin with, Figure 1 shows that during the first half of the sample period the public/private wage differentials for male and female workers remain relatively stable. Then, during the second half of the sample period, these differentials start to increase favouring the public sector. In addition, the public/private wage differential for male workers does not appear to be statistically different from that of female workers. Notice that the previous analysis was undertaken at a very high level of aggregation, in the sense that we only looked at average wages in the public and private sectors by gender. However, the advantage of using survey data (as we do in this document) is that other additional dimensions can be exploited as well. In particular, we also calculate:

- Average wage in the public and private sectors by city. Here we use the main seven metropolitan areas of the country that are available throughout the sample period; namely (in alphabetical order) Bucaramanga, Barranquilla, Bogotá, Cali, Medellín, Manizales and Pasto.
- 2. Average wage in the public and private sectors by gender and by city.
- 3. Average wage in the public and private sectors by occupational category. Here we consider the following four categories: managerial, professional, office and others.
- 4. Average wage in the public and private sectors by city and by occupational category.

Figure 2 examines the regional dimension. This figure shows that the public/private sector wage differential seems to exhibit a similar pattern across cities. In addition, Pasto presents the highest wage differential throughout the sample period, although during the last part of the sample period the gap with Bucaramanga appears to be closing. This finding may be explained by the fact that Pasto exhibits a lower average wage in the private sector compared to other cities where smaller gaps are observed (as in the case of Bogotá). When one examines the gaps for male and female workers it is clear that the one for Pasto can be explained by the fact that the corresponding gap for men is wider than that observed for women; see Figures 3 and 4.

Figures 5 to 8 plot wage gaps by occupational position. These figures illustrate an interesting finding. While the public/private wage differentials for managerial and professional employees (i.e. white collar employees) do not exhibit a significant variation, for the other two occupational positions (office and others) the wage differentials show a slight increase.

Figures 9 to 22 summarise wage differentials with respect to Bogotá. When using Bogotá as the category to which others are compared, the computed gaps do not show large changes over the sample period. Here there are two, somewhat expected, results that must be highlighted: (i.) wages in Bogotá are higher than in other cities; and (ii.) in the private sector the wage gaps with respect to Bogotá are much larger than those observed for the public sector.

Figure 23 compares the distribution of wages for the years 1985 and 2005. As can be seen from the figure, the distribution of wages have not changed considerable through time. An interesting finding is that private sector wages appear more concentrated around the mean than public sector ones. When the data are further analysed by occupational categories, other interesting results emerge. The distribution for managerial workers is very similar in the private and public sectors. For office workers the distribution of wages tends to be more concentrated, while for professionals and workers in managerial activities the wage distributions exhibit more variation around their respective means. On average, public sector wages tend to be higher than private sector ones.

Figure 24 displays the distribution of wages in the public and private sectors by city. In general terms, the patterns discussed in the previous paragraph can also be observed when examining wage distributions across cities, perhaps with the exception of Pasto, where wages are found to be less concentrated. Medellín is the city with the largest concentration in wage distribution, and the largest mean in the private sector for 2005.

The previous results describe the behaviour of wages over time. However, for a better understanding of the size of wage differentials is necessary to consider aspects such as human capital formation and demographic (market segmentation) factors. Thus, human capital formation appears in wage differentials when looking at the occupational profile of an individual, while demographic factors appear when examining differences by gender or by city. In fact, interesting results are uncovered when one relates wage differentials to age (as a proxy of an individual's work experience), years of schooling, gender and occupational category.

In the case of the age, our results indicate that men have more experience for all occupational positions, although differences in experience are smaller for blue collar employees. By sector, higher levels of experience are found in public sector workers, although in the case of white collar employees these differences tend to be smaller. Important differences in experience are also found within blue collar employees, which suggests that being an "office employee" or being another type of "blue collar employee" is determined by the educational level of the individual. The results just described provide the necessary justification for studying the relationship between public and private wages at a more disaggregated level.

Thus far, the analysis of wage differentials has been based on an informal visual inspection of the resulting series. In the following section we provide a more formal statistical analysis to assess the time-series properties of the variables under consideration.

## 5 Empirical analysis

Fluctuations in private and public sector wages have the potential to influence labour mobility from the public to the private sector and viceversa. Against this background, an assessment of whether wages in these two sectors maintain a stable long-run equilibrium relationship can have implications for the necessity to adjust remuneration in the public sector. Hence, there is considerable value in understanding how public and private sector wages behave in relation to each other over time. Our analysis examines three different aspects of public/private sector wages. The first one is related to the time-series properties of the public/private wage differential (Tables 1 and 2). The second one is concerned with the behaviour of public sector wages relative to that existing in Bogotá for an individual with similar characteristics (Tables 3 and 4). Lastly, the third one is related to the behaviour of private sector wages relative to that existing in Bogotá for an individual with similar characteristics (Tables 5 and 6). Thus, this sort of analysis allows us to examine wage differentials existing between the public and private sectors, as well as those that arise within each of these two sectors.

For the purposes of our empirical analysis, we examine the time-series properties of the wage differentials calculated in the previous section. The conventional ADF is applied to examine the unit-root null hypothesis for each wage differential. The test is performed including an intercept in the test regression. Also, one lag of the dependent variable is included in the test regression in order to ensure that the residuals do not suffer from serial correlation; the number of lags of the dependent variables is kept to a minimum due to the relatively small number of observations (that is T = 22). In addition to the ADF test, we also apply the KPSS stationarity test. For this, the model with an intercept (and no trend) is adopted in the empirical analysis, which implies that we would be testing the null hypothesis that a wage differential is stationary around a level. As indicated in the previous section, the long-run variance required to calculate the KPSS statistic is consistently estimated using the new boundary condition rule put forward by Sul, Phillips, and Choi (2005). Furthermore, to correct for possible serial correlation the autoregressive processes in (14) are estimated for up to p = 3 lags, and the optimal number of lags is chosen based on the GETS algorithm. This algorithm involves testing whether the last autoregressive coefficient is statistically different from zero (say, at the 10% significance level); if it is not statistically significant, then the order of the autoregression is reduced by one until the last coefficient is statistically significant.

Then, we turn our attention to panel unit-root and panel stationarity tests. The main motivation for statistical testing using a panel of data instead of individual time series is that it has been noted that the power of the tests increases with the number of cross-sections in the panel. Thus, we first calculate the IPS test which tests the null hypothesis of a unit root in all individuals in the panel, and is based on the assumption of independence across individuals. Then, to allow for potential cross section dependence, we also calculate two additional tests: (i) the CIPS test for panel unit-root; and (ii.) the bootstrap Hadri panel stationarity test (the corresponding bootstrap p-values are based on 2,000 replications used to derive the empirical distributions of the test statistics). It should be recalled that failure to account for potential cross section dependence will result in severe size distortion of both the IPS and Hadri test statistics.

Table 1 reports the results of applying the ADF unit-root and KPSS stationarity tests to public/private wage differentials by population group (i.e. male, female and total) and city (i.e. Bucaramanga, Barranquilla, Bogotá, Cali, Medellín, Manizales and Pasto). The results of the ADF test consistently fail to reject the null hypothesis of non-stationarity, while for the KPSS test evidence is mixed (the null hypothesis of stationarity is rejected in 10 out of 21 possible cases). Thus, evidence that wages are linked by a stable long-run equilibrium relationship is not particularly strong, when the series are examined in isolation. The bottom part of Table 1 reports the results of applying the panel tests. According to our findings, we fail to reject the null hypothesis of joint non-stationarity when using the IPS test, but not when using the CIPS test. This result highlights the importance of allowing for the presence of cross section dependence in the data. Turning to the bootstrap Hadri test, we fail to reject the null hypothesis of joint stationarity for male workers. In the case of all (total) and female workers, the null hypothesis of joint stationarity is rejected at the 5% (but not at the 1%) significance level; the resulting test statistics for all and female workers are 3.718 [*p*-value = 0.021] and 4.532 [*p*-value = 0.026], respectively. The reason why the CIPS and Hadri tests offer conflicting results may be due to the fact that the wage differentials are rather aggregate, as they do not discriminate by occupational category.

Table 2 reports the results for public/private wage differential by occupational category and city. When the series are considered in isolation results tend to be conflicting in the sense that both the ADF and KPSS tests either reject or fail to reject the null hypothesis. Panel tests, on the other hand, tell a completely different story. In this case, the CIPS test rejects the null hypothesis of joint non-stationarity for all occupational categories (i.e. managerial, professional, office and other), suggesting that at least one of the series in each panel is stationary. Stronger evidence in favour of stationarity is provided by the bootstrap Hadri test, as we fail to reject the null hypothesis of joint stationarity for any of the four occupational categories under consideration. These findings suggest that public and private sector wages maintain a long-run equilibrium relationship when analysed by occupational category, and after taking into account the presence of cross section dependence in the form of shocks (or innovations) that affect all series simultaneously.

Table 3 shows that the results when the tests are applied to public sector wages relative to the wage of an individual in the public sector in Bogotá. The conventional ADF test rejects the existence of a unit root in the wage differentials under consideration, except in the cases of male workers in the cities of Cali and Medellín. As to the KPSS test, we reject the stationarity null hypothesis in the cases of female workers in the cities of Bucaramanga, Medellín and Manizales. For the remaining 13 out of 18 cases the results of the univariate tests are consistent, in the sense that the ADF test rejects the null hypothesis, but the KPSS does not. Regarding the panel tests, the CIPS fails to reject the unit-root null hypothesis, although the calculated test statistics (-2.082, -2.035 and -2.149 for total, male and female workers, respectively) are relatively close to the 10% critical value tabulated by Pesaran (2007) (i.e.

-2.210). Overall, the results support the view that wage differentials are stationary.

Table 4 presents the results for the case of the public sector wage differential for the four occupational categories that are considered, with respect to the public sector wage of an individual in the same occupational category in Bogotá. As can be seen from the table, the results of the univariate unit-root and stationarity tests provide tend to favour the view that the wage differentials are stationary, in particular when looking at the results of the KPSS test (where we fail to reject the null of stationarity in 20 out of 24 cases). Turning to the results of the panel tests, they also provide support that the public sector wage differential with respect to Bogotá are stationary, with the exception of the CIPS test for individuals whose occupational category is "office".

The last two tables relate to wage differentials that involve the private sector. Table 5 shows the results when the tests are applied to private sector wages with respect to the wage of an individual with similar characteristics in the private sector in Bogotá. In this case the results of the univariate ADF and KPSS tests provide mixed evidence. However, the panel IPS, CIPS and Hadri tests provide support for the view that private sector wage differentials with respect to Bogotá are stationary when considered as a panel of data. Indeed, while the IPS and CIPS reject the unit-root null hypothesis for the total population as well as for male and female workers, the Hadri tests fail to reject the null of stationarity.

Table 6 reports the results of private sector wage differentials with respect to Bogota for the occupational categories under consideration. In this case the results of the univariate tests are mixed. The panel tests, on the other hand, provide support for the view that the differentials are stationary, especially after allowing for the effect of cross section dependence (the CIPS test statistic for "professional" workers is close to the 10% critical value).

## 6 Concluding remarks

In this document we have examined the time-series properties of the wage differentials that arise between the public and private sector in Colombia. The analysis has been based on information taken from nationwide household surveys. The utilisation of survey data offers the advantage that one can go beyond the typical calculation of the public/private sector wage differential for male and female workers, as well as for all workers. Indeed, the data can be analysed at a very high level of disaggregation. In particular, the dimensions that were studied in this document include: (i.) gender; (ii.) regional; and (iii.) occupational category.

Our findings indicate conflicting results in the unit-root and stationarity tests when one focuses on wage differentials at an aggregate level (such as for men, women and both). This is regardless of whether one is looking at each wage differential in isolation of the others, or jointly as a panel of data.

However, when we analyse wage differentials at higher levels of disaggregation, treat them jointly as a panel of data, and allow for the presence of potential cross section dependence, there is more supportive evidence for the view that wage differentials are stationary. In other words, average wages (in levels) appear to be linked by a stable long-run equilibrium relationship. In graphical terms, this long-run equilibrium relationship can be depicted as a 45 degree line, which plays the role of an attractor around which average wages fluctuate; in the short run, it may well be that average wages lie above (or below) this attractor. The economic importance of this finding is that despite the fact that wage differentials exist, they have not been consistently increasing (or decreasing) over time. If average wages were not linked by a long-run equilibrium relationship, there would exist non-negligible incentives for workers to move out of the public sector, which would undoubtedly jeopardise the ability of the public sector to attract and retain a productive and competitive labour force.

Group	City	p	ADF	p	KPSS	
Total	Bucaramanga	1	-0.151	1	0.581	‡
	Barranquilla	1	-1.629	1	0.253	
	Bogotá	1	-0.488	2	0.144	
	Cali	1	0.884	1	0.524	‡
	Medellín	1	-1.007	1	0.334	
	Manizales	1	-0.709	1	0.371	t
	Pasto	1	-0.438	1	0.396	t
Male	Bucaramanga	1	-0.598	1	0.297	
	Barranquilla	1	-1.182	3	0.443	t
	Bogotá	1	-1.241	2	0.132	
	Cali	1	-0.024	1	0.308	
	Medellín	1	-1.102	2	0.194	
	Manizales	1	-0.778	1	0.402	t
	Pasto	1	-0.742	2	0.267	
Female	Bucaramanga	1	-0.234	1	1.076	‡
	Barranquilla	1	-2.082	1	0.484	‡
	Bogotá	1	-0.330	2	0.131	
	Cali	1	0.196	1	0.375	t
	Medellín	1	-1.114	1	0.380	†
	Manizales	1	-0.877	1	0.253	
	Pasto	1	-0.859	1	0.205	

Table 1. public/private wage differential Univariate unit-root and stationarity tests

9	TDC	GIDG		TT 1 .	
Group	IPS	CIPS		Hadrı	
Total	2.795	-3.040	‡	3.718	‡
Male	1.954	-3.490	‡	2.218	
Female	2.099	-2.930	‡	4.532	‡

Notes: † and ‡ indicate 10 and 5% levels of significance, respectively, based on critical values calculated from the response surfaces in MacKinnon (1991) for the ADF test, and Sephton (1995) for the KPSS test. The IPS statistic is compared against the (lower tail of the) standard normal distribution, while the CIPS statistic is compared against critical values tabulated by Pesaran (2007). To account for potential cross section dependence, the statistical significance of the Hadri test is based on a bootstrap procedure that is implemented using 2,000 replications.

Occupational category	City	p	ADF		p	KPSS	
Managerial	Bucaramanga	1	-2.389		3	0.159	
	Barranquilla	1	-1.407		1	0.330	
	Bogotá	1	-0.910		2	0.057	
	Cali	1	-3.356	‡	1	0.582	‡
	Medellín	1	-2.640		1	0.357	†
	Manizales	1	-1.779		2	0.205	
	Pasto	1	-1.724		1	0.305	
Professional	Bucaramanga	1	-2.060		1	0.356	
	Barranguilla	1	-2.969	†	1	0.256	
	Bogotá	1	-1.964		1	0.314	
	Cali	1	-1.861		1	0.401	†
	Medellín	1	-2.382		1	0.429	†
	Manizales	1	-1.174		1	0.276	
	Pasto	1	-4.153	‡	1	0.415	†
Office	Bucaramanga	1	-0.812		2	0.156	
0	Barranquilla	1	-0.572		$\frac{-}{2}$	0.193	
	Bogotá	1	-0.737		1	0.201	
	Cali	1	-0.244		1	0.197	
	Medellín	1	-1.587		1	0.328	
	Manizales	1	-0.475		1	0.216	
	Pasto	1	0.428		†	0.254	
Other	Bucaramanga	1	-0.341		1	0.270	
Other	Barranguilla	1 1	-0.541 9.519		1	0.210 0.449	†
	Barranquina Borotá	1	-2.512		1 9	0.442 0.157	'
	Coli	1 1	0.004 0.011		2 1	0.137 0.241	
	Vadallín	1 1	0.011		1	0.241 0.256	
	Manizalos	1 1	-0.202		1 1	0.200	
	Pasto	1	-1.000		1	0.014 0.265	
	1 0000	Ŧ	1.050		T	0.200	

Table 2. public/private wage differential by occupational category and cityUnivariate unit-root and stationarity tests

Occupational category	IPS	(	CIPS		Hadri
Managerial	-1.421	† -	2.275	t	2.090
Professional	-2.352	‡ _;	3.015	‡	3.298
Office	2.613		3.060	‡	0.873
Other	2.025	-2	2.954	‡	1.950

	~						
Group	City	p	ADF		p	KPSS	
Total	Bucaramanga	1	-3.741	‡	1	0.223	
	Barranquilla	1	-3.048	‡	1	0.204	
	Cali	1	-2.648	†	1	0.223	
	Medellín	1	-2.690	†	1	0.118	
	Manizales	1	-3.885	‡	1	0.157	
	Pasto	1	-2.956	†	1	0.155	
Male	Bucaramanga	1	-4.441	‡	1	0.232	
	Barranquilla	1	-2.968	†	1	0.129	
	Cali	1	-2.478		1	0.264	
	Medellín	1	-1.843		2	0.098	
	Manizales	1	-2.698	†	1	0.158	
	Pasto	1	-2.780	†	1	0.237	
Female	Bucaramanga	1	-3.046	‡	1	0.390	†
	Barranquilla	1	-2.737	†	1	0.166	
	Cali	1	-3.012	†	1	0.125	
	Medellín	1	-3.356	‡	1	0.513	‡
	Manizales	1	-4.988	‡	1	0.813	‡
	Pasto	1	-3.051	‡	1	0.208	

Table 3. Public wage relative to public wage in BogotáUnivariate unit-root and stationarity tests

Group	IPS		CIPS	Hadri
Total	-4.214	ţ	-2.082	0.100
Male	-3.462	‡	-2.035	0.211
Female	-4.735	‡	-2.149	3.397

Occupational category	City	p	ADF		p	KPSS	
Managerial	Bucaramanga	1	-3.238	‡	1	0.237	
	Barranquilla	1	-2.281		1	0.347	
	Cali	1	-1.836		1	0.408	t
	Medellín	1	-3.050	‡	1	0.350	
	Manizales	1	-1.734		2	0.049	
	Pasto	1	-3.126	‡	1	0.310	
			1 100		0	0.070	
Professional	Bucaramanga	1	-1.496		3	0.058	Ť
	Barranquilla	1	-2.193	т	1	0.460	T
	Cali	1	-3.265	Ŧ	1	0.214	
	Medellín	1	-2.294	т	1	0.101	
	Manizales	1	-3.415	Ŧ	1	0.217	
	Pasto	1	-2.080		1	0.372	Ť
Office	Decomo mo on mo	1	4 491	t	1	0.100	
Onice	Ducaramanga	1	-4.421 1.010	т	1	0.100	
	Darranquina Cal:	1	-1.910	t	1	0.058 0.177	
	Call Madalléa	1	-3.779	+	1	0.110	
	Medellin	1	-3.701	+ +	1	0.118	
	Manizales	1	-2.075	+	1	0.069	
	Pasto	T	-2.987	'	1	0.090	
Other	Bucaramanga	1	-2.865	†	1	0.237	
	Barranguilla	1	-4.222	‡	3	0.825	‡
	Cali	1	-1.634		1	0.081	
	Medellín	1	-1.527		1	0.172	
	Manizales	1	-1.840		1	0.271	
	Pasto	1	-1.904		1	0.249	

 Table 4. Public wage relative to public wage in Bogotá by occupational category

 Univariate unit-root and stationarity tests

Occupational category	IPS		CIPS		Hadri
Managerial	-2.633	‡	-2.592	‡	1.906
Professional	-2.410	‡	-1.946		1.093
Office	-4.432	‡	-2.324	‡	-1.263
Other	-2.090	‡	-2.713	‡	2.298

Group	City	p	ADF		p	KPSS	
Total	Bucaramanga	1	-1.809		1	0.652	‡
	Barranquilla	1	-3.001	†	1	0.532	‡
	Cali	1	-1.561		1	0.331	
	Medellín	1	-2.392		1	0.223	
	Manizales	1	-2.525		1	0.417	†
	Pasto	1	-1.932		1	0.127	
Male	Bucaramanga	1	-1.784		1	0.480	†
	Barranquilla	1	-2.728	t	1	0.494	‡
	Cali	1	-1.756		1	0.268	
	Medellín	1	-2.036		1	0.176	
	Manizales	1	-2.204		1	0.229	
	Pasto	1	-2.276		1	0.136	
Female	Bucaramanga	1	-2.113		1	0.607	‡
	Barranquilla	1	-3.624	‡	1	0.408	†
	Cali	1	-1.686		2	0.296	
	Medellín	1	-3.674	‡	1	0.272	
	Manizales	1	-2.950	†	1	0.848	‡
	Pasto	1	-1.985		1	0.135	

Table 5. Private wage relative to private wage in BogotáUnivariate unit-root and stationarity tests

Group	IPS		CIPS		Hadri
Total	-1.760	‡	-2.394	‡	3.594
Male	-1.574	t	-2.389	‡	2.144
Female	-2.961	‡	-2.355	‡	4.416

Occupational category	City	p	ADF		p	KPSS	
Managerial	Bucaramanga	1	-1.793		1	0.227	
	Barranquilla	1	-0.946		2	0.093	
	Cali	1	-1.954		1	0.615	‡
	Medellín	1	-2.609		1	0.497	‡
	Manizales	1	-1.975		1	0.175	
	Pasto	1	-2.863	t	1	0.284	
	_			т			
Professional	Bucaramanga	1	-2.910	1	1	0.134	Ŧ
	Barranquilla	1	-2.154		1	0.623	+
	Cali	1	-2.212		1	0.503	‡
	Medellín	1	-2.426		1	0.297	
	Manizales	1	-2.330		1	0.649	‡
	Pasto	1	-2.203		2	0.127	
Office	Bucaramanga	1	-2.929	t	1	0.523	‡
	Barranquilla	1	-2.039		1	0.127	
	Cali	1	-1.718		1	0.428	†
	Medellín	1	-3.286	ţ	1	0.418	†
	Manizales	1	-2.214		1	0.273	
	Pasto	1	-2.403		1	0.160	
Other	Bucaramanga	1	-1.956		1	0.126	
	Barranquilla	1	-2.516		1	0.119	
	Cali	1	-2.144		1	0.087	
	Medellín	1	-2.005		1	0.102	
	Manizales	1	-1.746		1	0.151	
	Pasto	1	-1.131		1	0.081	

 Table 6. Private wage relative to private wage in Bogotá by occupational category

 Univariate unit-root and stationarity tests

Occupational category	IPS		CIPS		Hadri
Managerial	-1.299	t	-2.731	‡	2.456
Professional	-2.193	‡	-2.065		3.741
Office	-2.345	‡	-2.574	‡	2.570
Other	-1.025		-2.283	†	-1.103


Figure 1. Public/private sector wage differential by gender

Figure 2. Public/private sector wage differential by city





Figure 3. Male public/private sector wage differential by city

Figure 4. Female public/private sector wage differential by city





Figure 5. Office employees public/private sector wage differential by city

Figure 6. Other employees public/private sector wage differential by city





Figure 7. Managerial employees public/private sector wage differential by city

Figure 8. Professional employees public/private sector wage differential by city





Figure 9. Public sector wage differential with respect to Bogotá



Figure 10. Public sector wage differential with respect to Bogotá (male)

Figure 11. Public sector wage differential with respect to Bogotá (female)





Figure 12. Office employees public sector wage differential with respect to Bogotá

Figure 13. Other employees public sector wage differential with respect to Bogotá





Figure 14. Managerial employees public sector wage differential with respect to Bogotá

Figure 15. Professional employees public sector wage differential with respect to Bogotá





Figure 16. Private sector sector wage differential with respect to Bogotá  $\,$ 



Figure 17. Private sector wage differential with respect to Bogotá (male)

Figure 18. Private sector wage differential with respect to Bogotá (female)





Figure 19. Office employees private sector wage differential with respect to Bogotá

Figure 20. Other employees private sector wage differential with respect to Bogotá





Figure 21. Managerial employees private sector wage differential with respect to Bogotá

Figure 22. Professional employees private sector wage differential with respect to Bogotá







Figure 24. Distribution of wages by city



# References

- Adamchik, V. A. and A. S. Bedi (2000). Wage differentials between the public and the private sectors: Evidence from an economy in transition. *Labour Economics* 7, 203–224.
- Alvarez, L. J., J. Jareño, and M. Sebastian (1993). Salarios públicos, salarios privados e inflación dual. Banco de España Working Papers from Banco de España No 9320.
- Arango, L. E. and C. E. Posada (2002). Unemployment rate and the real wage behaviour: A neoclassical hint for the Colombian labour market adjustment. Applied Economics Letters 9, 425–428.
- Arango, L. E. and C. E. Posada (2007). Los salarios de los funcionarios públicos en colombia. Ensayos sobre Política Económica 55, 110–147.
- Banerjee, A. and M. Wagner (2009). Panel methods to test for unit roots and cointegration. In T. C. Mills and K. Patterson (Eds.), *Palgrave Handbook of Econometrics*. *Volume 2: Applied Econometrics*, pp. 632–726. Chippenham: Palgarve MacMillan.
- Bender, K. A. (2002). The central government-private sector wage differential. Journal of Economic Surveys 12, 177–220.
- Breitung, J. and M. H. Pesaran (2008). Unit roots and cointegration in panels. In L. Mátyás and P. Sevestre (Eds.), *The Econometrics of Panel Data*, pp. 279–322. Berlin: Springer-Verlag.
- Campbell, J. Y. and P. Perron (1991). Pitfalls and opportunities: What macroeconomists should know about unit roots. *NBER Macroeconomics Annual* 6, 141–201.
- Caner, M. and L. Kilian (2001). Size distortions of tests of the null hypothesis of stationarity: Evidence and implications for the PPP debate. *Journal of International Money and Finance 20*, 639–657.
- Carrión-i Silvestre, J. and A. Sansó (2006). A guide to the computation of stationarity tests. *Empirical Economics* 31, 433–448.
- Chang, Y. (2004). Bootstrap unit root tests in panels with cross-sectional dependency. Journal of Econometrics 120, 263–293.
- Dickey, D. A. and W. A. Fuller (1979). Distribution of the estimators for autoregressive time series with a unit root. Journal of the American Statistical Association 74, 427–431.
- Dickey, D. A. and W. A. Fuller (1981). Likelihood ratio statistics for autoregressive time series with a unit root. *Econometrica* 49, 1057–1072.
- Dustmann, C. and A. V. Soest (1998). Public and private sector wages of male workers in germany. *European Economic Review* 42, 1417–1441.
- Ehrenberg, R. G. and J. L. Schwarz (1986). Public sector labor markets. In O. Ashenfelter and R. Layard (Eds.), *Handbook of Labor Economics, Volume II*, pp. 1219–1267. Amsterdam: North-Holland.

- Franses, P. H. (1998). Time Series Models for Business and Economic Forecasting. Cambridge: Cambridge University Press.
- Fuller, W. (1976). Introduction to Statistical Time Series. New York, NY: Wiley.
- Galvis, L. A. (2010). Comportamiento de los salarios reales en colombia: Un análisis de convergencia condicional, 1984-2009. Documentos de Trabajo sobre Economía Regional. Banco de la República. No 127.
- Giulietti, M., J. Otero, and J. Smith (2009). Testing for stationarity in heterogeneous panel data in the presence of cross section dependence. *Journal of Statistical Computational and Simulation* 79, 195–203.
- Granger, C. W. J. and P. Newbold (1974). Spurious regressions in econometrics. Journal of Econometrics 2, 111–120.
- Gregory, R. and J. Borland (1999). Recent development in public sector labor market. In O. Ashenfelter and R. Layard (Eds.), *Handbook of Labor Economics, Volume*, pp. 3573–3630. Amsterdam: North Holland.
- Hadri, K. (2000). Testing for stationarity in heterogeneous panels. The Econometrics Journal 3, 148–161.
- Hadri, K. and R. Larsson (2005). Testing for stationarity in heterogeneous panel data where the time dimension is finite. *The Econometrics Journal* 8, 55–69.
- Hall, A. (1994). Testing for a unit root in time series with pretest data-based model selection. Journal of Business and Economic Statistics 12, 461–470.
- Hendry, D. F. (1980). Econometrics-alchemy or science? *Economica* 47, 387–406.
- Hundley, G. (1991). Public- and private-sector occupational pay structures. Industrial Relations: A Journal of Economy and Society 30, 417–434.
- Im, K., M. H. Pesaran, and Y. Shin (2003). Testing for unit roots in heterogeneous panels. *Journal of Econometrics* 115, 53–74.
- Iregui, A. M. (2005). Decentralised provision of quasi-private goods: The case of colombia. *Economic Modelling 22*, 683–706.
- Kwiatkowski, D., P. C. B. Phillips, P. Schmidt, and Y. Shin (1992). Testing the null hypothesis of stationarity against the alternative of a unit root. *Journal of Econometrics* 54, 159–178.
- Lamo, A., J. Perez, and L. Schuknecht (2008). Public and private sector wages. comovements and causality. European Central Bank. Working Paper Series. No 963.
- Lora, E. and G. Márquez (1998). The employment problem in Latin America: Perceptions and stylized facts. Inter- American Development Bank, Working Paper 371.
- MacKinnon, J. G. (1991). Critical values for cointegration tests. In R. F. Engle and C. W. J. Granger (Eds.), Long-Run Economic Relationships: Readings in Cointegration, pp. 267–276. Oxford: Oxford University Press.
- Maddala, G. S. and I. M. Kim (1998). Unit Roots, Cointegration, and Structural Change. Cambridge: Cambridge University Press.

- Maddala, G. S. and S. Wu (1999). A comparative study of unit root tests with panel data and a new simple test. Oxford Bulletin of Economics and Statistics 61, 631–652.
- Mueller, R. E. (1998). Public–private sector wage differentials in canada: Evidence from quantile regressions. *Economics Letters* 60, 229–235.
- Panizza, U. and C. Z.-W. Qiang (2005). Public–private wage differential and gender gap in latin america: Spoiled bureaucrats and exploited women? *Journal of Socio-Economics* 34, 810–833.
- Pederson, P. J., J. Schmidt-Sørensen, N. Smith, and N. Westergård-Nielsen (1990). Wage differentials between the public and private sectors. *Journal of Public Economics* 41, 125–145.
- Perron, P. (1988). Trends and random walks in macroeconomic time series : Further evidence from a new approach. Journal of Economic Dynamics and Control 12, 297–332.
- Pesaran, M. H. (2007). A simple panel unit root test in the presence of cross section dependence. *Journal of Applied Econometrics* 22, 265–312.
- Said, S. E. and D. A. Dickey (1984). Testing for unit roots in autoregressive-moving average models of unknown order. *Biometrika* 71, 599–607.
- Sephton, P. S. (1995). Response surface estimates of the KPSS stationarity test. Economics Letters 47, 255–261.
- Shapiro, C. and J. Stiglitz (1984). Equilibrium unemployment as a discipline device. American Economic Review 74, 433–444.
- Stelcner, M., J. V. der Gaag, and W. Vijverberg (1989). A swithching regression model of public.private sector wage differentials in peru: 1985-1986. The Journal of Human Resources 26, 545–559.
- Strauss, J. and T. Yigit (2003). Shortfalls of panel unit root testing. *Economics Letters* 81, 309–313.
- Sul, D., P. Phillips, and C. Choi (2005). Prewhitening bias in HAC estimation. Oxford Bulletin of Economics and Statistics 67, 517–546.
- Tansel, A. (1998). Public-private employment choice, wage differentials and gender in turkey. Working Papers from Economic Growth Center, Yale University.
- Yule, G. U. (1926). Why do we sometimes get nonsense correlations between time series? a study in sampling and the nature of time series. *Journal of the Royal Statistical Society 89*, 1–64.

Year	Р	ublic secto	or	Private sector				
	Total	Male	Female	Total	Male	Female		
1984	353,829	208,881	144,948	$1,\!627,\!085$	1,082,388	544,697		
1985	$325,\!880$	186,414	139,466	$1,\!583,\!657$	1,033,240	$550,\!417$		
1986	$339,\!190$	$197,\!353$	$141,\!837$	1,716,702	$1,\!118,\!180$	$598,\!522$		
1987	$334,\!863$	$195,\!259$	$139,\!604$	$1,\!813,\!819$	$1,\!170,\!527$	$643,\!292$		
1988	$373,\!542$	$216,\!328$	$157,\!214$	$1,\!959,\!645$	1,260,705	$698,\!940$		
1989	$384,\!937$	$213,\!866$	$171,\!071$	$2,\!073,\!134$	$1,\!311,\!360$	761,774		
1990	$408,\!453$	$230,\!556$	$177,\!897$	$2,\!226,\!004$	$1,\!407,\!493$	$818,\!511$		
1991	$408,\!637$	$227,\!033$	$181,\!604$	$2,\!202,\!571$	$1,\!349,\!841$	852,730		
1992	$381,\!678$	204,755	176,923	$2,\!345,\!323$	$1,\!431,\!521$	$913,\!802$		
1993	$333,\!928$	$178,\!987$	$154,\!941$	$2,\!430,\!904$	$1,\!464,\!311$	966, 593		
1994	$372,\!447$	$202,\!342$	$170,\!105$	2,706,833	$1,\!640,\!939$	$1,\!065,\!894$		
1995	$353,\!633$	$188,\!241$	$165,\!392$	$2,\!674,\!789$	$1,\!581,\!852$	$1,\!092,\!937$		
1996	$357,\!547$	$192,\!411$	$165,\!136$	$2,\!604,\!232$	$1,\!541,\!076$	$1,\!063,\!156$		
1997	$359{,}647$	$177,\!172$	$182,\!475$	$2,\!457,\!603$	$1,\!432,\!777$	1,024,826		
1998	$385,\!943$	$193,\!585$	$192,\!358$	$2,\!542,\!859$	$1,\!467,\!139$	1,075,720		
1999	$357,\!503$	$188,\!129$	$169,\!374$	$2,\!358,\!363$	$1,\!345,\!970$	$1,\!012,\!393$		
2000	$326,\!495$	$160,\!153$	$166,\!342$	$2,\!378,\!727$	$1,\!342,\!309$	$1,\!036,\!418$		
2001	$282,\!292$	140,793	$141,\!499$	$2,\!168,\!774$	$1,\!194,\!977$	$973,\!797$		
2002	$277,\!244$	$133,\!873$	$143,\!371$	$2,\!217,\!958$	$1,\!223,\!511$	$994,\!447$		
2003	$248,\!915$	129,764	$119,\!151$	$2,\!233,\!967$	$1,\!236,\!701$	$997,\!266$		
2004	$283,\!588$	$137,\!255$	$146,\!333$	$2,\!454,\!722$	$1,\!356,\!310$	$1,\!098,\!412$		
2005	$324,\!464$	$166,\!478$	$157,\!986$	$2,\!940,\!383$	$1,\!628,\!181$	$1,\!312,\!202$		

Appendix 1.1 Number of individuals

Notes: The number of individuals presented in this Appendix is calculated using the corresponding expansion factor in each survey.

Year			Ci	ty			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	$\mathbf{Pasto}$
1984	20,704	24,507	181,288	41,199	$58,\!343$	$14,\!650$	13,138
1985	$16,\!593$	$20,\!600$	$171,\!626$	$39,\!249$	$54,\!165$	12,088	$11,\!559$
1986	19,762	$22,\!530$	174,432	$41,\!481$	54,069	$15,\!106$	11,810
1987	22,757	$23,\!620$	166,775	$43,\!854$	$52,\!403$	$13,\!681$	11,773
1988	$28,\!536$	$23,\!392$	185,794	$43,\!014$	$62,\!179$	$17,\!303$	$13,\!324$
1989	$27,\!815$	30,288	187,822	49,866	55,008	$18,\!174$	$15,\!964$
1990	$28,\!683$	$38,\!237$	190,270	$48,\!146$	$66,\!677$	20,181	$16,\!259$
1991	28,765	$40,\!567$	189,885	$48,\!987$	65,759	19,468	$15,\!206$
1992	28,165	$31,\!008$	$176,\!659$	$45,\!529$	$66,\!885$	$18,\!450$	$14,\!982$
1993	$27,\!233$	26,727	$145,\!827$	$41,\!809$	60,951	$16,\!105$	$15,\!276$
1994	$25,\!390$	$25,\!522$	$175,\!696$	$45,\!190$	$67,\!504$	16,239	16,906
1995	$26,\!054$	$26,\!590$	$158,\!816$	$44,\!658$	$65,\!410$	$16,\!133$	$15,\!972$
1996	$26,\!367$	$33,\!150$	167, 121	$37,\!696$	59,736	$16,\!572$	$16,\!905$
1997	27,098	$35,\!244$	170,971	$36,\!321$	$57,\!908$	$15,\!152$	$16,\!953$
1998	27,743	$34,\!310$	181,472	$47,\!165$	$63,\!490$	$16,\!342$	$15,\!421$
1999	21,037	$32,\!117$	170,931	$44,\!611$	$59,\!523$	$15,\!813$	$13,\!471$
2000	$20,\!114$	$25,\!096$	161,795	41,717	$51,\!319$	13,067	$13,\!387$
2001	$20,\!444$	$28,\!153$	$135,\!385$	$31,\!066$	$43,\!167$	$12,\!225$	$11,\!852$
2002	$15,\!548$	26,402	$146,\!356$	$27,\!472$	38,030	$12,\!105$	$11,\!331$
2003	15,967	$23,\!006$	$123,\!699$	$25,\!262$	$39,\!430$	$11,\!472$	$10,\!079$
2004	$17,\!959$	$23,\!029$	150,581	$24,\!659$	45,031	12,036	$10,\!293$
2005	$21,\!407$	$26,\!464$	$162,\!027$	$35,\!814$	$56,\!191$	11,725	$10,\!836$

Appendix 1.2 (continued) Number of individuals in the public sector

Voor			Ci	+ +			
rear	Duconomon co	Downon quillo	Porretá	Coli	Modellín	Manizalog	Decto
1004	Ducaramanga	Darranquina			245 019	Manzales	
1984	84,973	135,610	748,996	240,995	345,813	42,152	28,546
1985	69,006	$113,\!673$	$777,\!391$	$241,\!494$	$317,\!379$	$35,\!575$	$29,\!139$
1986	80,274	$123,\!517$	820,343	$259,\!273$	$365,\!858$	40,206	$27,\!231$
1987	99,348	$134,\!515$	$832,\!096$	$291,\!914$	$382,\!537$	43,772	$29,\!637$
1988	$113,\!445$	$134,\!398$	884,786	$318,\!553$	$428,\!234$	$50,\!171$	30,058
1989	$119,\!812$	$167{,}548$	$958,\!686$	$323,\!317$	$421,\!620$	$51,\!382$	30,769
1990	$128,\!173$	$188,\!116$	1,044,400	$337,\!039$	439,076	57,062	$32,\!138$
1991	$140,\!582$	$192,\!307$	$995,\!893$	$343,\!222$	$443,\!070$	$53,\!819$	$33,\!678$
1992	142,964	206,746	$1,\!103,\!447$	$335,\!836$	$457,\!010$	60,496	38,824
1993	$150,\!386$	209,307	$1,\!117,\!665$	$365,\!417$	$492,\!807$	$59,\!243$	$36,\!079$
1994	164,222	226,787	$1,\!273,\!863$	$381,\!906$	$556,\!343$	$62,\!814$	40,898
1995	$168,\!085$	200,097	$1,\!304,\!696$	364,460	$531,\!846$	$62,\!981$	$42,\!624$
1996	$162,\!258$	$203,\!854$	$1,\!276,\!899$	$335,\!213$	$520,\!229$	$61,\!998$	43,781
1997	$155,\!079$	$203,\!107$	$1,\!151,\!012$	$337,\!285$	$500,\!584$	66,312	44,224
1998	149,741	$205,\!843$	$1,\!222,\!758$	355,767	$501,\!067$	$62,\!305$	$45,\!378$
1999	$126,\!499$	$186,\!355$	1,162,433	$346,\!872$	$440,\!373$	$55,\!119$	40,712
2000	$141,\!974$	$188,\!651$	$1,\!172,\!697$	359,314	$418,\!428$	$53,\!399$	44,264
2001	149,463	$195,\!476$	$1,\!033,\!847$	$331,\!969$	$354,\!741$	$59,\!905$	$43,\!373$
2002	$138,\!938$	$191,\!629$	$1,\!086,\!762$	$321,\!426$	$368,\!550$	68,902	41,751
2003	$150,\!117$	177,026	$1,\!052,\!160$	$338,\!429$	407,271	$63,\!866$	$45,\!098$
2004	$164,\!653$	$170,\!653$	$1,\!150,\!760$	$367,\!267$	490,928	$65,\!633$	44,828
2005	$198,\!533$	$211,\!382$	$1,\!386,\!525$	$454,\!267$	$575,\!506$	$66,\!556$	$47,\!614$

Appendix 1.3 (continued) Number of individuals in the private sector

Year			Ci	ty			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	11,837	15,719	105,325	25,982	34,073	8,204	7,741
1985	8,828	$11,\!964$	96,977	$23,\!902$	$31,\!548$	$6,\!479$	6,716
1986	$11,\!376$	$13,\!947$	98,735	$26,\!105$	$31,\!389$	$8,\!682$	$7,\!119$
1987	$13,\!173$	$15,\!287$	$95,\!498$	$26,\!885$	$29,\!529$	$7,\!896$	$6,\!991$
1988	$16,\!256$	14,017	108,533	26,024	$33,\!659$	9,787	$^{8,052}$
1989	$14,\!973$	17,502	$101,\!997$	$29,\!812$	$30,\!537$	9,544	9,501
1990	15,755	$23,\!168$	106,414	$28,\!239$	$36,\!216$	11,568	$9,\!196$
1991	$16,\!321$	24,237	$104,\!054$	$26,\!605$	$36,\!208$	10,772	8,836
1992	$14,\!937$	$18,\!148$	92,727	$24,\!609$	35,500	$10,\!252$	$8,\!582$
1993	14,716	15,796	74,002	$24,\!369$	$32,\!847$	$8,\!412$	$8,\!845$
1994	$13,\!538$	$13,\!959$	$94,\!054$	$25,\!605$	$36,\!805$	$8,\!629$	9,752
1995	14,008	$14,\!318$	84,322	$25,\!012$	$34,\!151$	$7,\!899$	8,531
1996	$13,\!341$	$18,\!521$	89,192	22,211	31,726	8,201	9,219
1997	$13,\!144$	$18,\!288$	78,442	$20,\!490$	30,214	$7,\!598$	$8,\!996$
1998	13,503	$17,\!590$	89,609	$25,\!266$	$31,\!099$	$7,\!866$	$^{8,652}$
1999	$11,\!385$	$17,\!052$	$89,\!689$	$23,\!352$	$31,\!619$	$7,\!690$	$7,\!342$
2000	$10,\!377$	$12,\!186$	77,791	$21,\!831$	$24,\!648$	$6,\!053$	7,267
2001	$9,\!458$	$13,\!531$	68,309	$15,\!842$	$21,\!050$	$6,\!137$	$6,\!466$
2002	6,919	12,369	69,591	$13,\!260$	$19,\!652$	6,260	$5,\!822$
2003	$7,\!181$	$11,\!575$	66,929	$12,\!878$	$19,\!908$	5,796	$5,\!497$
2004	9,089	10,522	$71,\!899$	$12,\!179$	22,161	$5,\!907$	$5,\!498$
2005	11,012	$13,\!255$	82,337	19,338	$28,\!982$	5,737	$5,\!817$

Appendix 1.4 (continued) Number of male individuals in the public sector

Year			С	lity			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	55,858	96,373	479,970	164,797	234,543	30,115	20,732
1985	42,086	79,025	$493,\!287$	160,798	$211,\!487$	$25,\!377$	$21,\!180$
1986	$52,\!309$	$86,\!171$	$515,\!145$	$170,\!876$	246,732	$27,\!357$	$19,\!590$
1987	$63,\!005$	$93,\!991$	$516,\!326$	$193,\!197$	$252,\!675$	$30,\!449$	20,884
1988	$72,\!468$	$93,\!996$	$548,\!011$	209,718	$279,\!471$	$35,\!967$	$21,\!074$
1989	$74,\!152$	111,700	$582,\!647$	$214,\!384$	271,721	$35,\!290$	$21,\!466$
1990	79,992	$127,\!549$	$640,\!875$	$217,\!665$	$279,\!260$	40,232	$21,\!920$
1991	$88,\!332$	$129,\!426$	584,102	$215,\!330$	$274,\!472$	36,210	$21,\!969$
1992	$85,\!413$	$139,\!099$	$645,\!553$	$209,\!117$	$284,\!813$	40,786	26,740
1993	88,983	$138,\!416$	$646,\!441$	$225,\!989$	$303,\!330$	$37,\!957$	$23,\!195$
1994	100,809	$151,\!675$	$738,\!948$	$235,\!985$	$344,\!973$	$42,\!356$	$26,\!193$
1995	103,928	129,752	733,731	219,760	$325,\!523$	$41,\!548$	$27,\!610$
1996	$96,\!986$	$131,\!016$	$726,\!827$	$202,\!893$	$315,\!382$	$41,\!459$	26,513
1997	90,072	$125,\!333$	$641,\!095$	$202,\!172$	$302,\!877$	$43,\!493$	27,735
1998	$87,\!389$	$127,\!177$	682,046	$208,\!616$	$293,\!151$	$39,\!876$	$28,\!884$
1999	70,086	$114,\!672$	641,208	$203,\!218$	$256,\!025$	34,721	26,040
2000	77,747	$115,\!095$	$643,\!559$	209,981	$237,\!883$	$31,\!601$	$26,\!443$
2001	85,102	$117,\!917$	$548,\!511$	190,326	$190,\!210$	36,324	$26,\!587$
2002	$78,\!397$	116,969	$571,\!251$	$182,\!327$	$206,\!207$	42,779	$25,\!581$
2003	82,791	108,565	560,270	$194,\!501$	$223,\!875$	38,724	$27,\!975$
2004	$93,\!667$	$105,\!240$	614,756	$208,\!846$	268,731	38,307	26,763
2005	$108,\!846$	129,286	740,947	260,847	$320,\!843$	$39,\!127$	$28,\!285$

Appendix 1.5 (continued) Number of male individuals in the private sector

Year			Ci	ty			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	8,867	8,788	$75,\!963$	15,217	24,270	6,446	$5,\!397$
1985	7,765	$8,\!636$	$74,\!649$	$15,\!347$	$22,\!617$	$5,\!609$	4,843
1986	$8,\!386$	8,583	$75,\!697$	$15,\!376$	$22,\!680$	$6,\!424$	$4,\!691$
1987	9,584	8,333	$71,\!277$	16,969	$22,\!874$	5,785	4,782
1988	12,280	$9,\!375$	$77,\!261$	$16,\!990$	$28,\!520$	7,516	$5,\!272$
1989	$12,\!842$	12,786	$85,\!825$	$20,\!054$	$24,\!471$	$8,\!630$	$6,\!463$
1990	12,928	15,069	$83,\!856$	19,907	30,461	$8,\!613$	7,063
1991	12,444	$16,\!330$	$85,\!831$	$22,\!382$	$29,\!551$	$8,\!696$	$6,\!370$
1992	$13,\!228$	$12,\!860$	$83,\!932$	20,920	$31,\!385$	$8,\!198$	6,400
1993	$12,\!517$	10,931	$71,\!825$	$17,\!440$	28,104	$7,\!693$	$6,\!431$
1994	$11,\!852$	11,563	$81,\!642$	$19,\!585$	$30,\!699$	7,610	$7,\!154$
1995	12,046	$12,\!272$	$74,\!494$	$19,\!646$	$31,\!259$	8,234	$7,\!441$
1996	13,026	$14,\!629$	$77,\!929$	$15,\!485$	$28,\!010$	$8,\!371$	$7,\!686$
1997	$13,\!954$	$16,\!956$	$92,\!529$	$15,\!831$	$27,\!694$	$7,\!554$	$7,\!957$
1998	$14,\!240$	16,720	$91,\!863$	$21,\!899$	$32,\!391$	$8,\!476$	6,769
1999	$9,\!652$	$15,\!065$	$81,\!242$	$21,\!259$	$27,\!904$	$8,\!123$	$6,\!129$
2000	9,737	$12,\!910$	84,004	$19,\!886$	$26,\!671$	7,014	$6,\!120$
2001	10,986	$14,\!622$	$67,\!076$	$15,\!224$	$22,\!117$	6,088	$5,\!386$
2002	$^{8,629}$	14,033	76,765	14,212	$18,\!378$	$5,\!845$	5,509
2003	8,786	$11,\!431$	56,770	$12,\!384$	19,522	$5,\!676$	$4,\!582$
2004	8,870	12,507	$78,\!682$	$12,\!480$	$22,\!870$	$6,\!129$	4,795
2005	$10,\!395$	13,209	$79,\!690$	$16,\!476$	$27,\!209$	$5,\!988$	$5,\!019$

Appendix 1.6 (continued) Number of female individuals in the public sector

Year			С	ity			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	29,115	39,237	269,026	$76,\!198$	111,270	12,037	7,814
1985	26,920	$34,\!648$	$284,\!104$	$80,\!696$	$105,\!892$	$10,\!198$	$7,\!959$
1986	27,965	$37,\!346$	$305,\!198$	$88,\!397$	$119,\!126$	$12,\!849$	$7,\!641$
1987	$36,\!343$	40,524	315,770	98,717	$129,\!862$	$13,\!323$	8,753
1988	40,977	40,402	336,775	$108,\!835$	148,763	14,204	$8,\!984$
1989	$45,\!660$	$55,\!848$	$376,\!039$	$108,\!933$	$149,\!899$	16,092	9,303
1990	48,181	60,567	$403,\!525$	$119,\!374$	$159,\!816$	$16,\!830$	10,218
1991	$52,\!250$	$62,\!881$	411,791	$127,\!892$	$168,\!598$	$17,\!609$	11,709
1992	$57,\!551$	$67,\!647$	$457,\!894$	126,719	$172,\!197$	19,710	$12,\!084$
1993	$61,\!403$	70,891	$471,\!224$	$139,\!428$	$189,\!477$	$21,\!286$	$12,\!884$
1994	$63,\!413$	$75,\!112$	$534,\!915$	$145,\!921$	$211,\!370$	$20,\!458$	14,705
1995	$64,\!157$	70,345	$570,\!965$	144,700	$206,\!323$	$21,\!433$	$15,\!014$
1996	$65,\!272$	$72,\!838$	$550,\!072$	$132,\!320$	$204,\!847$	$20,\!539$	$17,\!268$
1997	$65,\!007$	77,774	509,917	$135,\!113$	197,707	$22,\!819$	$16,\!489$
1998	$62,\!352$	$78,\!666$	540,712	$147,\!151$	$207,\!916$	$22,\!429$	$16,\!494$
1999	$56,\!413$	$71,\!683$	$521,\!225$	$143,\!654$	$184,\!348$	$20,\!398$	$14,\!672$
2000	$64,\!227$	$73,\!556$	$529,\!138$	$149,\!333$	$180,\!545$	21,798	$17,\!821$
2001	$64,\!361$	$77,\!559$	$485,\!336$	$141,\!643$	$164{,}531$	$23,\!581$	16,786
2002	$60,\!541$	$74,\!660$	$515,\!511$	$139,\!099$	$162,\!343$	$26,\!123$	$16,\!170$
2003	$67,\!326$	$68,\!461$	$491,\!890$	$143,\!928$	$183,\!396$	$25,\!142$	$17,\!123$
2004	70,986	$65,\!413$	$536,\!004$	$158,\!421$	$222,\!197$	$27,\!326$	$18,\!065$
2005	$89,\!687$	82,096	$645,\!578$	$193,\!420$	$254,\!663$	$27,\!429$	19,329

Appendix 1.7 (continued) Number of female individuals in the private sector

Year		Public sect	or			Private se	ctor	
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	7,575	116,510	104,078	$125,\!666$	21,174	123,783	300,559	1,181,569
1985	6,168	112,767	$93,\!139$	$113,\!806$	$20,\!641$	$130,\!454$	$281,\!311$	$1,\!151,\!251$
1986	$7,\!172$	$115,\!294$	$97,\!024$	119,700	$18,\!569$	$142,\!531$	300,503	$1,\!255,\!099$
1987	$6,\!899$	109,731	$95,\!803$	$122,\!430$	20,713	$142,\!686$	$307,\!089$	$1,\!343,\!331$
1988	$6,\!485$	130,203	$105,\!156$	$131,\!698$	$27,\!535$	$157,\!372$	$330,\!506$	$1,\!444,\!232$
1989	$9,\!438$	$135,\!188$	105,763	$134,\!548$	$33,\!109$	188,757	$353,\!351$	$1,\!497,\!917$
1990	9,072	$146,\!642$	$107,\!027$	145,712	$38,\!642$	$195,\!655$	$376,\!301$	$1,\!615,\!406$
1991	9,719	$144,\!855$	$107,\!602$	$146,\!461$	$34,\!635$	$201,\!535$	387,737	$1,\!578,\!664$
1992	$7,\!586$	$145,\!811$	$94,\!249$	$134,\!032$	$42,\!435$	$217,\!110$	406,244	$1,\!679,\!534$
1993	$7,\!295$	$123,\!157$	$83,\!575$	$119,\!901$	$44,\!651$	$219,\!275$	417,775	1,749,203
1994	9,787	$145,\!261$	$91,\!128$	$126,\!271$	$52,\!433$	$253,\!360$	$488,\!610$	$1,\!912,\!430$
1995	9,500	$135,\!672$	86,721	121,740	$43,\!297$	$253,\!834$	488,900	$1,\!888,\!758$
1996	11,329	$143,\!954$	80,868	$121,\!396$	49,079	$260,\!430$	$478,\!913$	$1,\!815,\!810$
1997	$12,\!257$	$154,\!027$	$81,\!456$	$111,\!907$	$51,\!975$	$287,\!339$	$475,\!463$	$1,\!642,\!826$
1998	$12,\!615$	170,945	$88,\!611$	113,772	60,804	$314,\!614$	$479,\!980$	$1,\!687,\!461$
1999	10,771	$153,\!852$	$78,\!417$	$114,\!463$	$60,\!657$	$277,\!559$	447,768	$1,\!572,\!379$
2000	$12,\!492$	$142,\!278$	$71,\!105$	$100,\!620$	60,575	$286,\!976$	448,709	$1,\!582,\!467$
2001	$12,\!306$	$128,\!608$	$56,\!952$	84,426	59,917	$257,\!654$	419,401	$1,\!431,\!802$
2002	$8,\!419$	$132,\!032$	$61,\!358$	$75,\!435$	$73,\!505$	$261,\!304$	$436,\!217$	$1,\!446,\!932$
2003	$8,\!591$	$123,\!402$	$48,\!897$	68,025	$81,\!837$	266,871	$428,\!926$	$1,\!456,\!333$
2004	$7,\!297$	144,764	$56,\!526$	$75,\!001$	94,804	$311,\!363$	$484,\!982$	$1,\!563,\!573$
2005	$14,\!812$	$160,\!541$	$64,\!052$	$85,\!059$	$123,\!155$	$354,\!460$	$589,\!886$	$1,\!872,\!882$

## Appendix 1.8 (continued) Number of individuals by occupational category

Year		Male				Female		
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	5,211	$58,\!532$	52,601	92,537	2,364	$57,\!978$	51,477	33,129
1985	4,576	54,760	44,289	82,789	1,592	58,007	48,850	$31,\!017$
1986	$5,\!591$	$56,\!648$	46,737	$88,\!377$	1,581	$58,\!646$	50,287	31,323
1987	$5,\!846$	$52,\!309$	46,188	90,916	1,053	$57,\!422$	$49,\!615$	$31,\!514$
1988	4,926	$62,\!610$	$51,\!595$	$97,\!197$	1,559	$67,\!593$	$53,\!561$	$34,\!501$
1989	$7,\!428$	$64,\!171$	$47,\!351$	94,916	2,010	71,017	$58,\!412$	$39,\!632$
1990	$5,\!853$	71,879	50,935	$101,\!889$	3,219	74,763	56,092	$43,\!823$
1991	$7,\!174$	69,074	49,276	$101,\!509$	2,545	75,781	58,326	44,952
1992	4,798	66,719	$40,\!359$	$92,\!879$	2,788	79,092	$53,\!890$	$41,\!153$
1993	$5,\!381$	$53,\!298$	37,761	$82,\!547$	1,914	69,859	$45,\!814$	$37,\!354$
1994	$5,\!987$	65,783	40,326	90,246	3,800	79,478	50,802	36,025
1995	$6,\!055$	59,242	$35,\!620$	$87,\!324$	$3,\!445$	$76,\!430$	51,101	34,416
1996	$7,\!199$	64,316	$33,\!539$	$87,\!357$	4,130	$79,\!638$	$47,\!329$	34,039
1997	$6,\!839$	62,443	$34,\!246$	$73,\!644$	5,418	$91,\!584$	47,210	38,263
1998	7,714	$73,\!349$	$33,\!243$	$79,\!279$	4,901	$97,\!596$	55,368	$34,\!493$
1999	6,203	66,169	$33,\!453$	$82,\!304$	4,568	$87,\!683$	44,964	$32,\!159$
2000	$7,\!895$	54,738	26,364	$71,\!156$	4,597	$87,\!540$	44,741	29,464
2001	$6,\!519$	$51,\!879$	$22,\!096$	60,299	5,787	76,729	$34,\!856$	$24,\!127$
2002	4,425	54,233	21,360	$53,\!855$	$3,\!994$	77,799	$39,\!998$	$21,\!580$
2003	$5,\!085$	54,920	$17,\!867$	$51,\!892$	3,506	$68,\!482$	31,030	$16,\!133$
2004	4,724	$56,\!678$	20,941	$54,\!912$	2,573	88,086	$35,\!585$	20,089
2005	$9,\!491$	64,475	26,912	$65,\!600$	5,321	96,066	$37,\!140$	$19,\!459$

# Appendix 1.9 (continued) Number of individuals in the public sector by occupational category

Year		Male				Female		
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	17,294	73,492	142,082	849,520	3,880	50,291	158,477	332,049
1985	$15,\!518$	76,750	$133,\!804$	$807,\!168$	$5,\!123$	53,704	$147,\!507$	$344,\!083$
1986	$13,\!855$	80,820	$138,\!264$	$885,\!241$	4,714	61,711	$162,\!239$	369,858
1987	$15,\!436$	$77,\!052$	$138,\!942$	$939,\!097$	$5,\!277$	$65,\!634$	$168,\!147$	404,234
1988	20,007	84,116	$150,\!307$	$1,\!006,\!275$	7,528	$73,\!256$	$180,\!199$	$437,\!957$
1989	$23,\!080$	$101,\!254$	$157,\!578$	1,029,448	10,029	$87,\!503$	195,773	468,469
1990	$26,\!670$	$106,\!188$	164,418	$1,\!110,\!217$	11,972	$89,\!467$	$211,\!883$	$505,\!189$
1991	$23,\!561$	$107,\!551$	$165,\!070$	$1,\!053,\!659$	$11,\!074$	$93,\!984$	$222,\!667$	$525,\!005$
1992	27,792	$115,\!525$	$175,\!118$	$1,\!113,\!086$	$14,\!643$	$101,\!585$	$231,\!126$	566,448
1993	27,014	$113,\!388$	176,775	$1,\!147,\!134$	$17,\!637$	$105,\!887$	$241,\!000$	602,069
1994	$31,\!457$	131,924	205,779	$1,\!271,\!779$	20,976	$121,\!436$	$282,\!831$	$640,\!651$
1995	24,088	$130,\!182$	$201,\!072$	$1,\!226,\!510$	19,209	$123,\!652$	$287,\!828$	662,248
1996	$25,\!554$	$134,\!620$	$202,\!954$	$1,\!177,\!948$	$23,\!525$	$125,\!810$	$275,\!959$	$637,\!862$
1997	$28,\!254$	$147,\!683$	200,531	$1,\!056,\!309$	23,721	$139,\!656$	$274,\!932$	586,517
1998	$34,\!134$	160, 184	$195,\!049$	1,077,772	$26,\!670$	$154,\!430$	284,931	$609,\!689$
1999	$32,\!270$	$133,\!241$	181,733	998,726	$28,\!387$	$144,\!318$	266,035	$573,\!653$
2000	$34,\!393$	$134,\!669$	182,063	$991,\!184$	$26,\!182$	$152,\!307$	$266,\!646$	$591,\!283$
2001	$35,\!130$	$122,\!948$	169,944	866,955	24,787	134,706	$249,\!457$	$564,\!847$
2002	40,082	120,953	$176,\!410$	886,066	$33,\!423$	$140,\!351$	$259,\!807$	560,866
2003	$45,\!437$	$125,\!423$	$181,\!334$	884,507	36,400	141,448	$247,\!592$	$571,\!826$
2004	$53,\!203$	149,704	$207,\!151$	$946,\!252$	41,601	$161,\!659$	$277,\!831$	$617,\!321$
2005	65.858	174.564	247.299	1,140,460	57.297	179.896	342.587	732,422

## Appendix 1.10 (continued) Number of individuals in the private sector by occupational category

Year	P	ublic see	ctor	Private sector			
	Total	Male	Female	Total	Male	Female	
1984	162	169	152	99	103	92	
1985	190	199	179	116	121	108	
1986	228	237	214	140	146	128	
1987	275	281	266	165	168	158	
1988	351	367	329	206	212	194	
1989	444	465	418	263	269	253	
1990	553	573	527	338	349	317	
1991	734	772	685	444	449	434	
1992	945	946	943	560	574	537	
1993	$1,\!272$	$1,\!309$	1,226	778	813	722	
1994	1,723	$1,\!834$	$1,\!587$	999	1,021	964	
1995	1,942	$1,\!997$	$1,\!879$	$1,\!136$	$1,\!174$	1,077	
1996	$2,\!465$	$2,\!536$	2,383	$1,\!371$	$1,\!405$	$1,\!317$	
1997	$3,\!077$	$3,\!149$	3,000	$1,\!666$	1,720	$1,\!583$	
1998	$3,\!916$	$4,\!052$	3,775	$2,\!005$	$2,\!051$	$1,\!936$	
1999	$4,\!376$	4,500	4,240	$2,\!172$	$2,\!176$	$2,\!167$	
2000	$4,\!900$	4,976	4,823	$2,\!343$	$2,\!358$	2,322	
2001	$5,\!382$	$5,\!237$	$5,\!531$	$2,\!285$	$2,\!270$	$2,\!304$	
2002	5,724	$5,\!619$	5,829	$2,\!448$	$2,\!376$	$2,\!548$	
2003	6,042	6,040	6,045	$2,\!536$	2,506	$2,\!576$	
2004	$6,\!170$	6,202	$6,\!139$	$2,\!820$	$2,\!815$	$2,\!826$	
2005	6,701	$6,\!621$	6,786	$3,\!031$	$3,\!013$	$3,\!056$	

#### Appendix 2.1 Average wage per-hour (Current pesos)

Notes: The average wages reported in this appendix are calculated using data from household surveys.

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	160	158	167	163	157	155	153
1985	187	190	199	190	186	174	183
1986	214	218	250	214	221	207	221
1987	259	251	300	279	264	244	272
1988	344	310	386	350	339	308	343
1989	426	410	495	442	419	432	411
1990	536	508	603	556	557	505	547
1991	698	655	833	711	764	648	689
1992	849	1,016	1,020	965	938	866	850
1993	$1,\!186$	$1,\!188$	1,362	$1,\!421$	$1,\!293$	$1,\!128$	1,219
1994	1,551	$1,\!613$	$1,\!995$	$1,\!645$	1,788	$1,\!445$	$1,\!490$
1995	1,990	1,911	2,142	2,022	1,725	1,778	$1,\!899$
1996	$2,\!593$	2,409	2,702	2,216	$2,\!376$	2,322	2,504
1997	$3,\!133$	3,002	$3,\!895$	2,714	3,022	$2,\!846$	$2,\!972$
1998	4,108	$3,\!800$	4,403	3,738	$3,\!991$	$3,\!680$	3,767
1999	4,223	4,249	$5,\!171$	4,240	4,275	$4,\!140$	$4,\!365$
2000	4,803	$5,\!280$	5,009	$4,\!480$	4,908	4,959	$4,\!817$
2001	$5,\!420$	$5,\!246$	6,034	$5,\!080$	$5,\!584$	$5,\!271$	$5,\!128$
2002	$5,\!488$	$5,\!531$	$6,\!537$	$5,\!488$	$5,\!533$	$5,\!277$	6,082
2003	$6,\!307$	5,756	$6,\!609$	$5,\!607$	$6,\!287$	$5,\!424$	$6,\!338$
2004	$6,\!382$	$5,\!479$	$6,\!699$	6,758	6,252	5,797	$6,\!148$
2005	$6,\!894$	$5,\!456$	7,724	7,025	6,788	6,202	6,747

### Appendix 2.2 (continued) Average wage per-hour in the public sector (Current pesos)

Year	r City						
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	87	100	108	102	94	88	62
1985	98	120	129	116	116	96	77
1986	120	143	161	141	131	113	87
1987	143	150	187	182	154	142	114
1988	178	182	236	217	199	183	140
1989	230	243	308	281	246	225	173
1990	296	313	400	351	323	278	221
1991	375	406	540	483	413	364	270
1992	472	525	664	607	531	474	352
1993	615	702	956	871	730	590	499
1994	812	1,004	1,218	999	897	785	635
1995	939	1,067	$1,\!395$	1,218	1,039	873	711
1996	1,207	1,362	$1,\!677$	$1,\!308$	$1,\!360$	1,213	1,018
1997	$1,\!448$	1,626	$2,\!479$	$1,\!640$	$1,\!641$	$1,\!423$	$1,\!182$
1998	1,730	2,006	2,823	$1,\!959$	$1,\!959$	1,754	$1,\!418$
1999	1,817	$2,\!179$	2,813	$2,\!321$	$2,\!051$	2,080	$1,\!587$
2000	2,057	2,372	$3,\!135$	2,261	$2,\!420$	2,085	$1,\!679$
2001	1,994	$2,\!437$	$2,\!857$	$2,\!384$	2,364	2,066	1,705
2002	2,061	$2,\!498$	3,312	$2,\!639$	$2,\!433$	2,060	$1,\!975$
2003	$2,\!148$	$2,\!498$	3,089	$2,\!596$	$2,\!841$	$2,\!336$	$2,\!000$
2004	2,363	$2,\!684$	$3,\!625$	2,920	3,034	2,533	2,128
2005	$2,\!661$	3,059	$3,\!897$	3,146	3,063	2,611	2,255

#### Appendix 2.3 (continued) Average wage per-hour in the private sector (Current pesos)

Year	City								
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto		
1984	173	165	175	170	160	158	162		
1985	210	200	209	195	188	177	191		
1986	227	233	264	217	227	217	229		
1987	265	259	306	274	270	247	295		
1988	373	326	401	363	346	327	364		
1989	442	439	516	467	430	449	442		
1990	565	548	616	583	560	507	581		
1991	729	719	866	777	800	653	733		
1992	868	897	1,044	985	970	859	869		
1993	1,239	1,260	1,410	$1,\!459$	1,269	$1,\!115$	1,316		
1994	$1,\!636$	1,557	$2,\!195$	$1,\!660$	1,948	1,509	$1,\!631$		
1995	2,149	1,991	2,145	$2,\!053$	1,734	1,779	2,030		
1996	2,589	2,393	2,802	2,231	$2,\!431$	2,369	2,750		
1997	$3,\!271$	$3,\!155$	3,999	$2,\!666$	3,046	$2,\!852$	$3,\!146$		
1998	4,336	$3,\!897$	4,494	3,724	$4,\!158$	3,869	$3,\!944$		
1999	4,092	4,420	5,543	4,097	4,214	4,366	$4,\!612$		
2000	4,742	5,234	4,839	4,420	$5,\!175$	$5,\!119$	$5,\!105$		
2001	5,216	5,326	$5,\!681$	$4,\!657$	5,332	$5,\!133$	$5,\!189$		
2002	5,368	5,564	6,242	4,825	$5,\!559$	5,223	6,119		
2003	6,169	5,517	6,476	$5,\!482$	$6,\!499$	$5,\!375$	6,593		
2004	$6,\!288$	5,508	6,738	6,410	6,330	5,926	6,232		
2005	6,569	$5,\!662$	7,354	$6,\!375$	7,048	5,966	7,024		

#### Appendix 2.4 (continued) Male average wage per-hour in the public sector (Current pesos)

Year	$\operatorname{City}$								
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto		
1984	91	105	113	106	97	86	63		
1985	101	126	135	115	122	96	78		
1986	127	153	170	145	137	116	87		
1987	145	155	194	184	159	142	111		
1988	187	189	248	222	205	182	138		
1989	234	257	313	291	251	223	173		
1990	304	322	423	368	332	277	220		
1991	384	417	550	486	425	357	267		
1992	499	547	681	634	545	466	352		
1993	641	724	1,037	913	758	583	486		
1994	831	1,055	1,256	1,026	916	768	629		
1995	970	1,093	$1,\!490$	$1,\!281$	1,068	863	697		
1996	1,243	1,402	1,753	$1,\!344$	$1,\!390$	1,220	1,030		
1997	1,505	1,702	2,726	$1,\!673$	$1,\!687$	$1,\!427$	$1,\!178$		
1998	1,775	2,068	$3,\!015$	2,041	2,022	1,735	1,364		
1999	1,793	2,221	2,883	2,313	2,073	2,076	$1,\!553$		
2000	2,129	2,393	3,328	2,242	2,365	$2,\!111$	1,584		
2001	1,990	2,401	$2,\!901$	2,418	2,412	2,046	$1,\!611$		
2002	2,102	2,506	$3,\!106$	$2,\!614$	$2,\!394$	2,015	$1,\!848$		
2003	$2,\!117$	2,499	3,121	$2,\!602$	2,842	2,299	$1,\!879$		
2004	2,368	2,703	3,711	$2,\!880$	$3,\!077$	2,565	2,008		
2005	2,679	3,049	3,939	3,128	3,097	2,532	2,141		

#### Appendix 2.5 (continued) Male average wage per-hour in the private sector (Current pesos)

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	144	145	157	150	153	151	141
1985	161	175	185	182	184	171	172
1986	196	195	232	208	213	194	209
1987	251	237	291	287	255	240	238
1988	308	287	366	329	329	284	311
1989	407	370	470	404	406	413	366
1990	501	446	585	518	554	503	502
1991	657	561	794	632	719	642	628
1992	828	$1,\!187$	994	941	902	875	825
1993	1,123	1,083	1,312	1,367	$1,\!322$	$1,\!142$	1,086
1994	$1,\!453$	$1,\!680$	1,762	$1,\!626$	1,588	$1,\!373$	$1,\!297$
1995	1,801	1,817	$2,\!138$	1,982	1,715	1,777	1,749
1996	2,596	$2,\!430$	2,585	$2,\!195$	$2,\!314$	$2,\!278$	2,208
1997	3,004	$2,\!837$	$3,\!806$	2,778	$2,\!994$	$2,\!839$	2,774
1998	$3,\!890$	$3,\!699$	4,313	3,756	$3,\!828$	3,503	$3,\!542$
1999	$4,\!379$	4,054	4,761	4,402	$4,\!345$	3,924	4,072
2000	4,871	$5,\!324$	$5,\!170$	4,549	$4,\!662$	4,820	$4,\!472$
2001	$5,\!596$	$5,\!170$	6,404	$5,\!534$	$5,\!829$	$5,\!411$	$5,\!053$
2002	$5,\!587$	5,501	6,806	$6,\!142$	5,506	$5,\!335$	6,043
2003	$6,\!423$	6,013	6,767	5,745	6,054	$5,\!475$	6,034
2004	6,477	$5,\!454$	$6,\!664$	$7,\!116$	$6,\!176$	$5,\!670$	$6,\!052$
2005	$7,\!237$	5,244	8,090	7,839	6,512	6,429	$6,\!421$

### Appendix 2.6 (continued) Female average wage per-hour in the public sector (Current pesos)

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	79	87	100	93	88	94	58
1985	92	107	117	116	102	95	75
1986	108	121	146	133	118	108	89
1987	140	139	174	178	145	140	121
1988	164	165	217	208	187	186	146
1989	222	215	299	262	236	229	173
1990	282	295	363	320	307	281	224
1991	359	384	524	478	393	379	276
1992	433	480	641	562	507	492	354
1993	578	659	844	801	683	602	523
1994	780	901	1,166	955	865	822	646
1995	889	1,019	1,271	$1,\!123$	991	891	736
1996	$1,\!152$	1,290	1,575	1,252	1,312	1,201	999
1997	1,368	1,505	2,165	1,590	1,569	1,416	$1,\!188$
1998	$1,\!666$	1,906	2,581	$1,\!843$	1,870	1,789	1,512
1999	1,849	2,112	2,726	2,331	2,022	2,088	$1,\!648$
2000	1,969	2,339	2,896	2,286	2,491	2,046	1,822
2001	1,999	2,493	2,806	2,338	2,309	2,097	1,857
2002	2,008	2,484	$3,\!544$	$2,\!672$	$2,\!483$	$2,\!134$	$2,\!173$
2003	$2,\!186$	2,497	3,051	2,588	2,840	2,392	$2,\!193$
2004	$2,\!355$	2,654	3,528	2,973	2,981	$2,\!488$	2,310
2005	2,639	3,076	3,848	$3,\!170$	3,021	2,724	2,423

#### Appendix 2.7 (continued) Female average wage per-hour in the private sector (Current pesos)

Year		Public sector	r		Private sector				
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other	
1984	374	242	139	98	425	239	105	80	
1985	426	273	160	122	433	276	120	95	
1986	466	327	197	147	507	330	143	116	
1987	695	407	233	173	681	386	170	137	
1988	828	524	297	213	852	489	207	170	
1989	992	657	367	269	1,029	617	258	216	
1990	$1,\!197$	819	453	337	1,300	747	331	278	
1991	$1,\!674$	1,088	591	440	1,751	977	438	364	
1992	2,069	1,406	740	563	2,228	$1,\!278$	550	453	
1993	$2,\!653$	1,883	1,030	743	$3,\!259$	2,014	720	608	
1994	$3,\!531$	2,501	$1,\!359$	994	$3,\!685$	$2,\!435$	959	781	
1995	$3,\!826$	2,777	$1,\!526$	$1,\!151$	$3,\!950$	$2,\!694$	$1,\!137$	901	
1996	$4,\!637$	$3,\!445$	$1,\!872$	$1,\!449$	5,013	$3,\!254$	$1,\!319$	$1,\!084$	
1997	$5,\!585$	4,310	$2,\!298$	1,785	5,272	$4,\!115$	$1,\!583$	$1,\!294$	
1998	7,925	$5,\!381$	$2,\!934$	2,079	$7,\!807$	4,770	1,916	1,505	
1999	8,870	$6,\!199$	$3,\!217$	2,265	7,118	4,978	$2,\!110$	$1,\!653$	
2000	9,519	6,784	$3,\!538$	2,562	8,333	6,038	2,231	$1,\!679$	
2001	9,446	$7,\!385$	4,136	2,710	7,801	$5,\!547$	$2,\!179$	$1,\!667$	
2002	9,422	7,725	4,225	2,927	8,535	5,746	$2,\!435$	1,771	
2003	$10,\!489$	$7,\!875$	4,536	$3,\!134$	$8,\!629$	5,703	$2,\!389$	$1,\!847$	
2004	10,913	7,979	$4,\!673$	$3,\!270$	9,030	$6,\!663$	$2,\!633$	2,021	
2005	11,627	8.476	5,076	3,627	9,231	6.843	2.770	2.226	

#### Appendix 2.8 (continued) Average wage per-hour by occupational category (Current pesos)

Appendix $2.9$ (continued)	
Average wage per-hour in the public sector by occupational c	ategory
(Current pesos)	

Year		Male				Female		
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	400	284	147	102	321	200	132	88
1985	444	316	169	127	371	234	151	108
1986	488	381	211	151	386	277	183	136
1987	731	465	238	179	465	356	228	156
1988	891	612	318	222	619	443	277	189
1989	1,038	750	390	285	833	574	347	233
1990	1,275	923	467	357	1,029	719	440	292
1991	1,705	1,258	608	471	1,573	933	574	368
1992	$2,\!059$	1,511	763	597	2,090	$1,\!317$	723	486
1993	2,793	2,139	$1,\!059$	796	2,205	$1,\!675$	1,006	617
1994	$3,\!345$	3,026	1,564	1,028	$3,\!896$	2,066	$1,\!193$	906
1995	$3,\!899$	$3,\!196$	$1,\!631$	$1,\!184$	3,668	$2,\!451$	$1,\!453$	1,070
1996	4,842	3,889	$1,\!945$	$1,\!489$	4,103	$3,\!086$	1,820	$1,\!352$
1997	$5,\!992$	4,758	2,516	$1,\!848$	4,800	$3,\!974$	$2,\!140$	$1,\!651$
1998	9,094	$6,\!126$	2,969	2,095	$5,\!603$	4,804	2,912	2,045
1999	10,261	6,944	$3,\!531$	$2,\!290$	6,415	$5,\!597$	$2,\!992$	2,208
2000	10,214	7,510	3,798	2,585	$7,\!846$	6,265	$3,\!378$	2,510
2001	10,002	$7,\!936$	4,328	$2,\!695$	8,560	$6,\!993$	4,016	2,749
2002	$8,\!618$	8,492	4,338	2,919	10,507	$7,\!176$	$4,\!155$	2,945
2003	$10,\!645$	8,726	4,934	$3,\!140$	10,166	$7,\!259$	4,269	$3,\!116$
2004	$12,\!146$	8,934	5,037	$3,\!258$	8,928	$7,\!350$	$4,\!438$	3,303
2005	$12,\!396$	9,248	5,312	$3,\!689$	10,211	$7,\!958$	4,904	$3,\!423$
Year		Male				Female		
------	------------	--------------	-----------	-----------	------------	--------------	-----------	-----------
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	440	275	107	84	360	187	103	71
1985	480	320	121	99	294	213	119	86
1986	549	401	146	121	371	236	141	104
1987	725	458	172	141	550	303	169	128
1988	941	587	205	176	605	383	208	155
1989	$1,\!109$	710	261	221	837	511	256	202
1990	$1,\!429$	870	331	289	1,010	602	331	252
1991	$1,\!898$	$1,\!137$	427	370	$1,\!442$	800	446	353
1992	$2,\!390$	$1,\!458$	551	469	1,905	1,077	549	420
1993	$3,\!651$	2,568	702	634	$2,\!642$	$1,\!442$	733	556
1994	4,034	2,795	944	812	$3,\!136$	2,047	969	715
1995	4,295	$3,\!196$	$1,\!165$	941	$3,\!472$	$2,\!157$	$1,\!118$	823
1996	$5,\!813$	3,823	$1,\!340$	$1,\!117$	3,906	$2,\!665$	$1,\!304$	1,020
1997	$5,\!834$	4,840	$1,\!563$	$1,\!360$	4,549	$3,\!345$	$1,\!599$	$1,\!160$
1998	8,736	5,565	$1,\!968$	1,558	6,559	$3,\!980$	$1,\!879$	$1,\!401$
1999	$7,\!904$	$5,\!636$	2,121	$1,\!699$	6,226	$4,\!375$	2,102	1,565
2000	8,909	$7,\!432$	$2,\!252$	1,706	7,571	$4,\!896$	2,216	$1,\!633$
2001	8,795	$6,\!118$	$2,\!086$	$1,\!697$	6,549	5,023	$2,\!246$	$1,\!616$
2002	9,731	$6,\!158$	2,211	1,794	$7,\!159$	$5,\!384$	$2,\!599$	1,731
2003	9,786	$6,\!487$	$2,\!256$	1,861	7,288	$5,\!040$	$2,\!488$	1,822
2004	10,093	$7,\!692$	$2,\!489$	$2,\!052$	$7,\!653$	5,792	2,741	$1,\!970$
2005	10,233	7,711	$2,\!617$	$2,\!249$	$8,\!130$	6,041	$2,\!883$	$2,\!189$

#### Appendix 2.10 (continued) Average wage per-hour in the private sector by occupational category (Current pesos)

Year			Mana	gerial			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	184	474	440	371	479	271	335
1985	379	355	544	379	362	250	287
1986	438	710	519	470	405	390	398
1987	514	629	813	698	643	521	470
1988	717	1,075	835	1,214	732	746	786
1989	881	$1,\!340$	$1,\!115$	934	683	929	888
1990	1,109	1,084	1,415	$1,\!316$	$1,\!121$	$1,\!110$	872
1991	$1,\!451$	$1,\!831$	1,987	$2,\!186$	$1,\!254$	1,418	$1,\!453$
1992	2,203	1,513	$2,\!344$	$2,\!061$	$2,\!196$	1,791	$1,\!532$
1993	$1,\!874$	$1,\!648$	$2,\!839$	2,784	6,021	1,815	2,038
1994	2,705	$2,\!664$	$5,\!400$	2,772	$3,\!001$	3,004	$2,\!371$
1995	3,776	$5,\!609$	4,221	4,946	$2,\!998$	3,515	$3,\!270$
1996	4,771	$3,\!523$	4,910	$3,\!981$	4,747	$5,\!681$	4,208
1997	5,269	$4,\!375$	7,508	$5,\!443$	$5,\!549$	6,503	4,865
1998	6,005	10,913	8,815	9,019	7,522	$13,\!965$	6,788
1999	$6,\!459$	$6,\!876$	14,706	$9,\!116$	8,799	$7,\!883$	$7,\!620$
2000	7,909	$29,\!167$	11,368	8,404	$^{8,131}$	20,041	$7,\!153$
2001	8,088	10,783	$13,\!918$	$8,\!296$	8,045	8,382	$9,\!057$
2002	7,739	8,529	$14,\!655$	$11,\!919$	7,731	8,969	$9,\!193$
2003	$11,\!158$	$8,\!384$	$15,\!952$	9,019	$10,\!353$	9,352	9,728
2004	12,211	8,010	$18,\!187$	$12,\!489$	$11,\!488$	7,910	10,811
2005	9,702	10,559	17,884	11,502	11,839	9,582	10,922

#### Appendix 2.11 (continued) Average wage per-hour in the public sector by occupational category (Current pesos)

Year	r Managerial							
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto	
1984	292	637	447	463	393	317	243	
1985	337	268	477	425	437	355	358	
1986	323	942	550	514	468	360	400	
1987	571	663	745	866	520	634	382	
1988	632	$1,\!150$	856	958	888	977	585	
1989	715	948	$1,\!181$	990	964	1,168	695	
1990	1,231	$1,\!315$	$1,\!421$	1,203	$1,\!258$	$1,\!359$	725	
1991	$1,\!424$	2,061	2,028	$1,\!676$	1,506	$2,\!187$	810	
1992	1,539	$2,\!542$	2,512	$1,\!951$	$2,\!154$	$1,\!889$	$1,\!575$	
1993	$1,\!661$	$2,\!867$	$3,\!651$	4,141	$3,\!367$	$2,\!375$	$1,\!926$	
1994	1,992	$2,\!863$	$5,\!336$	2,567	$3,\!412$	$3,\!524$	$2,\!236$	
1995	2,142	$3,\!627$	5,732	$4,\!148$	$3,\!111$	3,760	$2,\!553$	
1996	$3,\!092$	1,900	$5,\!812$	4,726	5,921	6,348	3,742	
1997	4,001	$2,\!562$	$7,\!253$	$5,\!841$	$4,\!437$	$7,\!417$	$3,\!444$	
1998	$4,\!692$	3,709	$11,\!415$	$7,\!119$	6,788	8,400	6,769	
1999	$4,\!129$	$5,\!371$	10,302	7,020	$5,\!898$	8,012	$5,\!485$	
2000	$5,\!170$	7,778	$11,\!235$	8,214	$8,\!553$	6,529	$6,\!258$	
2001	5,067	7,025	$11,\!908$	$7,\!831$	6,906	$8,\!987$	$7,\!088$	
2002	$5,\!929$	$6,\!644$	13,791	8,505	$^{8,218}$	$7,\!823$	$6,\!841$	
2003	$5,\!660$	7,041	$12,\!233$	$8,\!380$	9,349	$8,\!662$	$7,\!266$	
2004	6,064	$6,\!603$	$13,\!144$	9,053	9,502	9,254	6,021	
2005	6,916	7,080	14,331	10,015	8,702	7,693	6,086	

#### Appendix 2.12 (continued) Average wage per-hour in the private sector by occupational category (Current pesos)

Year			Profess	sional			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	232	247	252	261	224	223	228
1985	246	258	298	289	264	247	260
1986	290	338	359	343	309	287	308
1987	363	383	445	436	384	373	405
1988	475	445	575	548	516	481	507
1989	599	629	738	686	612	638	592
1990	757	773	916	870	803	740	770
1991	981	984	1,312	1,091	1,094	939	954
1992	1,237	1,903	1,485	$1,\!433$	1,382	1,228	$1,\!154$
1993	$1,\!657$	1,721	2,083	2,735	1,800	1,763	$1,\!626$
1994	$2,\!156$	2,355	2,790	$2,\!658$	2,718	$2,\!199$	2,056
1995	2,841	$2,\!680$	$3,\!179$	2,937	2,475	$2,\!642$	2,526
1996	3,416	3,238	4,075	3,759	3,312	3,336	$3,\!159$
1997	4,353	$3,\!849$	5,395	4,372	4,398	4,353	$3,\!866$
1998	$5,\!871$	4,802	6,458	5,748	$5,\!589$	$5,\!114$	4,727
1999	$6,\!189$	5,783	$7,\!606$	6,585	6,064	6,023	$5,\!691$
2000	6,399	6,888	7,333	$7,\!149$	6,954	6,929	6,260
2001	$7,\!669$	6,836	8,562	$7,\!407$	8,063	$7,\!617$	$6,\!409$
2002	8,226	7,113	8,785	7,791	8,065	$7,\!533$	7,273
2003	8,081	7,376	8,885	$8,\!399$	8,360	7,200	$7,\!574$
2004	$8,\!287$	6,666	9,367	9,043	8,020	$8,\!285$	$7,\!051$
2005	9,135	6,572	10,042	9,169	8,363	8,542	8,009

#### Appendix 2.13 (continued) Average wage per-hour in the public sector by occupational category (Current pesos)

Year			Profess	sional			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	172	219	252	236	241	222	206
1985	226	238	293	282	287	276	210
1986	250	313	347	321	363	297	206
1987	301	316	421	422	360	390	346
1988	373	362	564	513	473	456	325
1989	489	490	702	635	603	509	459
1990	648	559	876	768	772	678	552
1991	800	730	$1,\!186$	994	959	894	615
1992	1,051	1,005	$1,\!494$	1,244	1,311	1,234	869
1993	1,383	1,460	2,817	$1,\!846$	1,725	1,704	1,064
1994	$1,\!879$	2,165	$3,\!017$	2,132	2,232	$1,\!959$	$1,\!474$
1995	$2,\!193$	2,033	3,207	2,967	$2,\!576$	2,400	$1,\!657$
1996	2,862	2,510	$3,\!875$	3,006	$3,\!497$	3,323	2,763
1997	$3,\!519$	$3,\!158$	5,469	4,313	$4,\!393$	$3,\!670$	$2,\!877$
1998	4,383	$3,\!895$	6,207	4,616	4,711	4,512	$3,\!590$
1999	4,168	4,208	6,005	$5,\!456$	4,859	$5,\!317$	3,712
2000	$6,\!393$	5,034	$8,\!138$	4,864	$6,\!277$	4,810	5,033
2001	$5,\!177$	$5,\!113$	6,816	$5,\!164$	6,054	5,169	$4,\!391$
2002	4,560	4,737	7,852	$6,\!142$	$5,\!683$	$5,\!043$	4,925
2003	4,944	4,593	6,967	$5,\!848$	6,787	4,875	$4,\!899$
2004	$5,\!687$	4,766	8,999	6,935	$7,\!304$	$5,\!400$	$5,\!314$
2005	6,171	5,406	8,367	7,304	7,327	6,016	5,861

#### Appendix 2.14 (continued) Average wage per-hour in the private sector by occupational category (Current pesos)

Year			Offi	ce			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	126	144	147	131	129	143	119
1985	149	174	168	151	151	165	140
1986	189	197	208	196	193	175	181
1987	216	210	252	252	226	215	204
1988	291	277	326	312	277	265	253
1989	342	352	390	397	367	363	308
1990	457	454	477	456	451	430	409
1991	549	594	657	608	575	527	520
1992	669	719	799	782	796	715	621
1993	987	1,030	1,047	1,088	$1,\!194$	951	830
1994	1,226	1,318	$1,\!679$	1,274	1,296	1,121	1,006
1995	1,453	1,572	1,764	$1,\!611$	$1,\!430$	1,426	$1,\!134$
1996	1,988	2,274	1,909	1,715	1,936	$1,\!637$	$1,\!635$
1997	2,223	2,661	2,599	2,238	2,412	2,075	2,000
1998	3,026	3,551	2,748	2,784	2,914	2,895	$2,\!675$
1999	2,789	3,845	3,382	$3,\!170$	3,204	3,376	2,733
2000	3,503	4,119	3,292	3,277	3,545	3,964	3,019
2001	3,940	4,451	4,099	4,282	$4,\!127$	4,536	$3,\!458$
2002	3,592	4,029	5,584	4,020	$3,\!931$	4,179	3,778
2003	4,773	3,967	5,231	4,039	$4,\!350$	$4,\!671$	4,111
2004	4,782	4,598	4,796	4,950	4,575	4,499	4,762
2005	4,819	4,431	6,307	5,535	4,826	4,555	4,978

#### Appendix 2.15 (continued) Average wage per-hour in the public sector by occupational category (Current pesos)

Year			Off	.ce			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	90	111	112	102	97	93	75
1985	101	122	132	112	118	102	86
1986	124	142	159	139	136	132	109
1987	144	157	187	180	164	155	139
1988	181	188	224	216	200	215	164
1989	219	253	283	266	249	263	193
1990	287	319	369	324	330	305	250
1991	383	422	503	438	415	394	319
1992	472	529	619	545	538	500	437
1993	625	680	806	761	702	673	537
1994	853	968	1,088	939	882	846	664
1995	882	1,070	1,397	1,084	1,060	937	779
1996	1,136	1,514	1,500	$1,\!250$	1,290	1,172	983
1997	1,395	1,708	1,955	1,525	1,616	1,406	$1,\!189$
1998	1,714	2,221	2,202	1,795	1,907	1,736	1,409
1999	1,764	2,259	2,560	2,087	2,099	1,975	1,644
2000	1,938	2,729	2,503	2,017	2,179	$2,\!173$	1,717
2001	1,974	2,325	2,429	2,161	2,157	2,211	1,787
2002	2,093	2,390	3,284	2,430	2,292	2,169	1,951
2003	2,163	2,190	2,860	$2,\!387$	$2,\!480$	2,362	2,013
2004	2,336	2,466	$3,\!188$	$2,\!688$	2,731	2,364	$2,\!193$
2005	2,496	2.673	3.386	2.795	2,758	2,491	2.207

#### Appendix 2.16 (continued) Average wage per-hour in the private sector by occupational category (Current pesos)

Year			Oth	ner			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	$\mathbf{Pasto}$
1984	106	106	97	101	99	88	88
1985	129	144	118	127	127	97	103
1986	145	143	157	143	155	121	128
1987	170	191	174	183	171	155	150
1988	227	223	213	223	216	178	194
1989	261	277	269	276	283	258	247
1990	341	356	333	347	346	298	314
1991	450	463	436	435	474	384	388
1992	598	570	567	600	551	495	536
1993	774	783	739	815	699	639	738
1994	1,037	1,020	1,007	1,037	1,009	925	848
1995	1,235	$1,\!179$	1,177	$1,\!241$	1,114	976	1,073
1996	1,613	1,556	1,468	1,329	1,474	1,342	1,308
1997	1,796	2,082	2,129	1,731	1,743	1,610	1,535
1998	2,197	2,333	2,099	1,986	2,115	1,862	2,026
1999	2,313	2,257	2,465	2,161	2,337	2,111	2,264
2000	3,031	2,497	2,443	$2,\!431$	2,777	2,492	$2,\!354$
2001	2,662	$2,\!642$	2,551	2,755	2,809	2,894	2,583
2002	2,989	2,760	2,827	$3,\!276$	3,077	2,758	2,930
2003	$3,\!241$	2,896	$3,\!056$	$3,\!358$	$3,\!252$	3,026	$3,\!170$
2004	$3,\!646$	3,163	2,919	$3,\!853$	$3,\!371$	2,962	$3,\!355$
2005	3,785	3,202	3,791	3,858	3,961	3,292	3,345

#### Appendix 2.17 (continued) Average wage per-hour in the public sector by occupational category (Current pesos)

Year	r Other							
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto	
1984	76	84	82	83	81	73	51	
1985	82	105	99	94	101	76	65	
1986	104	117	128	119	115	92	75	
1987	125	128	146	152	138	115	94	
1988	154	161	180	180	175	155	116	
1989	202	209	232	231	216	188	142	
1990	256	273	313	290	275	233	179	
1991	333	347	418	401	351	303	233	
1992	422	440	493	508	451	392	291	
1993	543	567	668	703	609	479	390	
1994	705	803	875	839	736	645	489	
1995	826	908	1,030	971	864	688	568	
1996	1,013	1,111	1,257	1,074	1,104	942	735	
1997	1,181	1,329	1,790	1,248	$1,\!341$	1,114	891	
1998	1,399	1,546	1,857	1,559	1,546	$1,\!386$	1,060	
1999	1,500	1,735	1,944	1,739	$1,\!644$	1,589	1,248	
2000	1,501	1,743	1,957	1,711	1,789	$1,\!672$	1,220	
2001	1,540	1,821	1,804	1,824	1,800	1,590	1,217	
2002	1,710	1,935	1,963	1,910	1,834	1,600	1,416	
2003	1,728	1,946	1,964	1,971	1,977	1,791	$1,\!454$	
2004	1,885	2,152	2,180	2,095	$2,\!151$	1,974	1,560	
2005	2,077	2,362	2,465	2,368	2,330	2,034	1,706	

#### Appendix 2.18 (continued) Average wage per-hour in the private sector by occupational category (Current pesos)

Year	P	ublic see	ctor	Private sector				
	Total	Male	Female	Total	Male	Female		
1984	$5,\!195$	5,413	4,878	$3,\!187$	3,300	2,955		
1985	4,927	$5,\!146$	$4,\!634$	3,009	$3,\!122$	2,787		
1986	4,956	5,167	$4,\!655$	3,039	$3,\!170$	2,782		
1987	$4,\!850$	4,962	$4,\!690$	$2,\!906$	$2,\!970$	2,782		
1988	4,839	5,062	4,532	$2,\!836$	2,921	$2,\!676$		
1989	4,862	$5,\!088$	4,572	$2,\!880$	$2,\!940$	2,773		
1990	$4,\!690$	4,859	4,467	2,863	$2,\!959$	$2,\!689$		
1991	4,772	5,021	$4,\!450$	2,884	2,920	2,823		
1992	4,835	$4,\!841$	4,829	2,866	$2,\!936$	2,750		
1993	5,316	$5,\!474$	$5,\!124$	$3,\!251$	$3,\!398$	$3,\!017$		
1994	5,862	6,242	$5,\!399$	$3,\!399$	$3,\!473$	$3,\!280$		
1995	$5,\!466$	$5,\!620$	$5,\!289$	$3,\!197$	3,303	3,032		
1996	5,744	$5,\!910$	$5,\!553$	$3,\!194$	$3,\!274$	3,069		
1997	$6,\!051$	$6,\!193$	$5,\!901$	$3,\!276$	$3,\!382$	$3,\!113$		
1998	$6,\!490$	6,716	$6,\!256$	3,322	$3,\!398$	3,208		
1999	$6,\!541$	6,726	$6,\!337$	$3,\!247$	$3,\!252$	$3,\!239$		
2000	6,706	$6,\!809$	$6,\!601$	$3,\!206$	$3,\!227$	$3,\!177$		
2001	$6,\!822$	$6,\!638$	7,011	$2,\!896$	$2,\!878$	2,920		
2002	$6,\!823$	$6,\!696$	6,947	2,917	$2,\!832$	3,036		
2003	6,722	6,720	6,725	$2,\!821$	2,788	2,866		
2004	$6,\!482$	$6,\!515$	$6,\!449$	$2,\!962$	$2,\!958$	$2,\!968$		
2005	6,701	$6,\!621$	6,786	3,031	$3,\!013$	$3,\!056$		

#### Appendix 3.1 Average wage per-hour (2005 Pesos)

Notes: The average real wages reported in this appendix are calculated using the data from Appendix 1, deflated using the consumer price index 2005=100.

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	5,144	5,064	5,364	5,222	$5,\!050$	4,967	4,909
1985	4,846	4,907	$5,\!140$	4,921	$4,\!819$	4,501	4,743
1986	$4,\!650$	4,751	$5,\!444$	$4,\!647$	4,820	4,504	4,806
1987	4,576	$4,\!435$	$5,\!291$	4,918	$4,\!653$	4,311	4,803
1988	4,736	4,275	5,324	4,824	$4,\!667$	4,247	4,720
1989	4,660	4,484	$5,\!422$	4,833	$4,\!591$	4,730	4,504
1990	4,547	4,303	5,109	4,716	4,725	4,283	$4,\!633$
1991	4,536	4,260	5,418	$4,\!624$	4,966	4,211	$4,\!477$
1992	4,346	5,202	5,223	4,940	4,803	$4,\!432$	$4,\!351$
1993	4,957	4,966	$5,\!692$	$5,\!940$	$5,\!404$	4,716	$5,\!097$
1994	$5,\!277$	$5,\!488$	6,789	$5,\!598$	$6,\!085$	4,918	$5,\!070$
1995	$5,\!601$	$5,\!378$	6,028	$5,\!690$	4,855	$5,\!005$	$5,\!346$
1996	6,041	$5,\!614$	$6,\!295$	5,164	$5,\!537$	$5,\!411$	$5,\!835$
1997	6,162	5,904	$7,\!660$	$5,\!339$	5,944	$5,\!597$	$5,\!845$
1998	6,808	$6,\!297$	$7,\!296$	$6,\!196$	$6,\!614$	6,099	6,243
1999	6,312	$6,\!351$	7,730	6,338	$6,\!390$	$6,\!188$	6,524
2000	$6,\!573$	7,226	6,855	$6,\!131$	6,717	6,787	$6,\!592$
2001	$6,\!870$	$6,\!649$	$7,\!648$	$6,\!438$	7,078	$6,\!682$	$6,\!499$
2002	$6,\!541$	$6,\!592$	7,791	$6,\!541$	$6,\!595$	$6,\!290$	$7,\!249$
2003	7,016	6,403	$7,\!353$	6,238	6,994	6,034	$7,\!051$
2004	6,705	5,755	7,037	7,099	6,568	6,089	$6,\!459$
2005	$6,\!894$	$5,\!456$	7,724	7,025	6,788	6,202	6,747

#### Appendix 3.2 (continued) Average wage per-hour in the public sector (2005 Pesos)

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	2,782	3,196	3,470	3,276	3,012	2,825	1,984
1985	2,526	$3,\!107$	3,330	2,990	$2,\!990$	$2,\!474$	$1,\!995$
1986	$2,\!611$	$3,\!115$	$3,\!498$	$3,\!072$	$2,\!849$	$2,\!470$	1,902
1987	2,531	$2,\!655$	$3,\!296$	3,211	2,718	2,501	2,008
1988	$2,\!455$	2,504	3,251	$2,\!995$	2,742	2,518	1,930
1989	2,513	$2,\!658$	$3,\!372$	$3,\!073$	$2,\!689$	$2,\!464$	$1,\!893$
1990	2,509	$2,\!656$	$3,\!386$	2,974	2,737	$2,\!356$	1,872
1991	$2,\!435$	$2,\!643$	3,508	$3,\!142$	$2,\!686$	2,368	1,758
1992	$2,\!417$	$2,\!689$	$3,\!400$	$3,\!105$	2,717	$2,\!428$	1,804
1993	2,571	2,936	$3,\!997$	$3,\!641$	$3,\!050$	$2,\!467$	2,086
1994	2,761	3,415	4,146	$3,\!400$	$3,\!052$	$2,\!673$	2,162
1995	$2,\!643$	3,004	3,925	$3,\!429$	2,923	$2,\!456$	$2,\!001$
1996	2,811	$3,\!173$	$3,\!907$	3,048	$3,\!168$	2,827	$2,\!372$
1997	2,848	$3,\!199$	4,876	3,226	$3,\!227$	2,799	2,324
1998	2,867	3,324	$4,\!679$	$3,\!247$	$3,\!247$	$2,\!907$	$2,\!350$
1999	2,716	$3,\!257$	4,205	3,469	3,066	$3,\!110$	$2,\!372$
2000	2,815	$3,\!247$	4,290	$3,\!094$	$3,\!312$	2,853	2,298
2001	2,527	3,089	$3,\!621$	3,021	$2,\!996$	$2,\!619$	2,161
2002	2,456	2,977	3,948	$3,\!145$	2,900	$2,\!455$	$2,\!354$
2003	2,390	2,779	$3,\!436$	2,888	3,160	2,599	2,225
2004	$2,\!482$	2,820	3,808	$3,\!067$	$3,\!187$	$2,\!661$	2,236
2005	$2,\!661$	3,059	$3,\!897$	$3,\!146$	3,063	$2,\!611$	2,255

#### Appendix 3.3 (continued) Average wage per-hour in the private sector (2005 Pesos)

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	$5,\!536$	5,297	$5,\!607$	$5,\!456$	$5,\!139$	5,070	5,184
1985	$5,\!440$	$5,\!173$	5,410	$5,\!050$	4,852	4,569	$4,\!950$
1986	4,943	5,069	5,740	4,723	4,941	4,718	4,979
1987	$4,\!686$	4,572	$5,\!406$	$4,\!831$	4,768	$4,\!367$	$5,\!210$
1988	$5,\!134$	4,492	$5,\!520$	$5,\!008$	4,772	4,507	$5,\!015$
1989	4,837	4,806	$5,\!654$	$5,\!108$	4,708	4,918	4,836
1990	4,792	$4,\!645$	5,223	4,943	4,748	4,299	4,924
1991	4,738	$4,\!672$	$5,\!627$	$5,\!050$	5,202	4,243	4,764
1992	$4,\!442$	4,589	5,346	5,044	4,966	$4,\!395$	$4,\!446$
1993	$5,\!179$	5,267	$5,\!893$	6,101	$5,\!304$	$4,\!663$	$5,\!501$
1994	5,568	$5,\!299$	7,470	$5,\!648$	$6,\!629$	$5,\!133$	$5,\!550$
1995	6,049	$5,\!605$	6,037	5,778	$4,\!880$	5,008	5,714
1996	6,033	$5,\!576$	6,529	$5,\!197$	$5,\!663$	$5,\!519$	$6,\!407$
1997	$6,\!433$	6,204	7,864	$5,\!244$	$5,\!991$	$5,\!609$	$6,\!188$
1998	$7,\!186$	$6,\!458$	7,448	$6,\!171$	$6,\!891$	$6,\!412$	$6,\!536$
1999	$6,\!116$	$6,\!607$	8,286	$6,\!123$	$6,\!298$	$6,\!526$	$6,\!893$
2000	$6,\!489$	7,163	$6,\!622$	6,049	$7,\!083$	7,006	$6,\!986$
2001	$6,\!612$	6,751	7,201	$5,\!903$	6,759	6,507	$6,\!578$
2002	$6,\!398$	$6,\!631$	$7,\!440$	5,750	$6,\!625$	6,224	$7,\!293$
2003	6,863	$6,\!138$	7,204	6,098	$7,\!231$	$5,\!980$	$7,\!335$
2004	$6,\!605$	5,787	7,078	6,734	$6,\!649$	$6,\!225$	$6,\!547$
2005	6,569	$5,\!662$	$7,\!354$	$6,\!375$	7,048	5,966	7,024

 $\begin{array}{c} \mbox{Appendix 3.4 (continued}) \\ \mbox{Male average wage per-hour in the public sector} \\ \mbox{(2005 Pesos)} \end{array}$ 

Year	$\operatorname{City}$								
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto		
1984	2,909	3,365	$3,\!619$	3,411	3,101	2,749	2,034		
1985	$2,\!621$	$3,\!250$	3,503	2,984	3,160	$2,\!479$	2,015		
1986	2,754	$3,\!325$	$3,\!691$	3,161	$2,\!978$	2,529	$1,\!891$		
1987	2,565	2,739	$3,\!429$	$3,\!248$	2,798	2,511	$1,\!953$		
1988	2,572	$2,\!604$	$3,\!410$	3,062	$2,\!823$	2,501	$1,\!897$		
1989	2,564	2,812	$3,\!432$	$3,\!180$	2,746	$2,\!445$	$1,\!892$		
1990	2,580	2,729	$3,\!581$	$3,\!117$	$2,\!813$	$2,\!346$	1,861		
1991	2,496	2,712	$3,\!578$	$3,\!163$	2,765	2,321	1,739		
1992	$2,\!553$	2,802	$3,\!483$	$3,\!243$	2,789	$2,\!386$	1,800		
1993	$2,\!679$	3,027	4,336	$3,\!818$	3,168	$2,\!439$	2,030		
1994	2,829	$3,\!589$	4,275	$3,\!492$	$3,\!116$	$2,\!612$	2,142		
1995	2,730	3,077	4,194	$3,\!605$	$3,\!005$	$2,\!429$	1,962		
1996	$2,\!897$	3,267	4,085	$3,\!132$	3,239	$2,\!842$	$2,\!400$		
1997	2,961	$3,\!347$	5,362	$3,\!291$	$3,\!317$	$2,\!806$	$2,\!317$		
1998	2,941	$3,\!426$	4,997	$3,\!382$	$3,\!350$	$2,\!875$	2,261		
1999	$2,\!679$	3,319	4,309	$3,\!457$	3,098	$3,\!102$	$2,\!321$		
2000	2,913	$3,\!275$	4,554	3,069	3,236	$2,\!890$	2,168		
2001	2,523	3,043	$3,\!678$	$3,\!064$	$3,\!057$	$2,\!594$	2,042		
2002	2,505	2,987	3,702	$3,\!116$	2,853	2,401	2,202		
2003	$2,\!355$	2,780	$3,\!473$	$2,\!894$	$3,\!161$	$2,\!558$	2,090		
2004	$2,\!488$	$2,\!840$	$3,\!898$	3,026	3,232	$2,\!695$	2,109		
2005	2,679	3,049	3,939	3,128	3,097	2,532	2,141		

#### Appendix 3.5 (continued) Male average wage per-hour in the private sector (2005 Pesos)

Year	City									
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto			
1984	4,619	4,652	5,023	4,823	4,924	4,838	4,515			
1985	4,164	4,540	4,783	4,718	4,771	4,421	$4,\!457$			
1986	4,265	4,239	5,054	4,518	$4,\!646$	4,219	4,546			
1987	4,434	4,188	$5,\!133$	$5,\!059$	4,503	4,234	4,203			
1988	4,244	3,953	5,043	4,535	$4,\!536$	$3,\!914$	4,287			
1989	$4,\!457$	4,052	$5,\!139$	$4,\!423$	$4,\!441$	4,523	4,002			
1990	4,247	3,777	4,961	$4,\!389$	$4,\!697$	4,262	4,252			
1991	4,271	$3,\!647$	5,164	4,109	$4,\!675$	4,172	4,080			
1992	4,240	6,073	5,086	4,814	$4,\!616$	4,478	4,223			
1993	$4,\!696$	4,526	$5,\!483$	5,716	$5,\!525$	4,773	$4,\!541$			
1994	4,944	5,718	5,996	$5,\!532$	$5,\!405$	$4,\!671$	4,414			
1995	$5,\!070$	$5,\!113$	6,017	$5,\!578$	4,828	5,002	4,923			
1996	6,048	$5,\!662$	6,024	$5,\!115$	$5,\!391$	$5,\!307$	$5,\!146$			
1997	$5,\!907$	$5,\!580$	$7,\!485$	$5,\!464$	5,888	$5,\!584$	$5,\!456$			
1998	$6,\!447$	$6,\!130$	$7,\!148$	$6,\!224$	$6,\!344$	$5,\!805$	$5,\!870$			
1999	$6,\!545$	6,060	$7,\!117$	$6,\!580$	$6,\!494$	5,865	$6,\!087$			
2000	$6,\!666$	7,286	7,075	$6,\!225$	$6,\!380$	$6,\!597$	$6,\!121$			
2001	7,093	6,553	$8,\!117$	7,015	$7,\!388$	$6,\!859$	$6,\!405$			
2002	$6,\!659$	$6,\!556$	8,111	$7,\!321$	6,562	$6,\!359$	$7,\!203$			
2003	$7,\!146$	$6,\!690$	7,528	$6,\!391$	6,735	6,091	6,713			
2004	$6,\!804$	5,729	7,000	$7,\!475$	$6,\!487$	$5,\!957$	$6,\!358$			
2005	7,237	5,244	8,090	7,839	6,512	6,429	6,421			

#### Appendix 3.6 (continued) Female average wage per-hour in the public sector (2005 Pesos)

Year			Cit	y			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	2,540	2,784	3,200	2,985	2,815	$3,\!015$	1,852
1985	$2,\!380$	2,781	3,026	$3,\!004$	$2,\!633$	$2,\!462$	1,942
1986	$2,\!350$	$2,\!628$	$3,\!172$	$2,\!898$	2,570	$2,\!344$	1,928
1987	2,472	2,461	$3,\!076$	$3,\!140$	$2,\!554$	$2,\!477$	$2,\!134$
1988	$2,\!255$	2,274	2,994	2,866	$2,\!583$	2,563	2,005
1989	$2,\!434$	$2,\!353$	$3,\!278$	2,863	$2,\!582$	2,506	$1,\!896$
1990	$2,\!390$	2,503	3,074	2,712	$2,\!599$	$2,\!380$	$1,\!896$
1991	2,333	2,499	$3,\!408$	$3,\!106$	$2,\!555$	$2,\!465$	1,795
1992	2,216	$2,\!456$	3,282	$2,\!876$	$2,\!594$	2,518	1,813
1993	$2,\!415$	2,756	3,527	$3,\!350$	$2,\!857$	2,517	$2,\!186$
1994	$2,\!654$	3,064	3,967	$3,\!249$	2,944	2,798	$2,\!199$
1995	2,501	2,869	$3,\!576$	$3,\!160$	2,790	2,508	2,072
1996	$2,\!685$	3,006	$3,\!669$	2,916	$3,\!057$	2,797	2,329
1997	$2,\!690$	2,959	4,258	$3,\!127$	3,086	2,785	$2,\!336$
1998	2,762	$3,\!159$	4,278	$3,\!055$	3,099	2,965	2,506
1999	2,763	$3,\!157$	4,075	$3,\!485$	3,022	$3,\!122$	2,463
2000	$2,\!695$	3,202	3,963	$3,\!129$	$3,\!409$	2,800	$2,\!493$
2001	2,533	3,160	$3,\!557$	2,963	2,926	$2,\!658$	$2,\!353$
2002	2,393	2,960	4,224	$3,\!184$	2,959	2,544	$2,\!590$
2003	$2,\!432$	2,778	$3,\!394$	$2,\!879$	$3,\!159$	$2,\!661$	$2,\!440$
2004	2,474	2,788	3,706	$3,\!123$	$3,\!132$	$2,\!613$	$2,\!426$
2005	2,639	3,076	3,848	$3,\!170$	3,021	2,724	2,423

#### Appendix 3.7 (continued) Female average wage per-hour in the private sector (2005 Pesos)

3.7	Dublic costor				Drivete sector				
Year		Public secto	r	_		Private secto	or		
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other	
1984	$11,\!994$	7,756	$4,\!472$	$3,\!153$	$13,\!645$	$7,\!666$	$3,\!361$	2,577	
1985	11,010	7,067	$4,\!135$	$3,\!155$	11,203	$7,\!138$	$3,\!097$	2,461	
1986	$10,\!139$	$7,\!122$	4,281	$3,\!198$	11,029	$7,\!174$	$3,\!119$	2,520	
1987	12,263	7,190	4,116	3,048	12,023	6,813	3,008	$2,\!421$	
1988	11,410	7,221	4,091	2,939	11,740	6,733	2,850	2,342	
1989	10,857	$7,\!196$	4,014	2,949	11,258	6,749	2,825	2,360	
1990	$10,\!145$	6,938	$3,\!842$	2,857	11,020	6,331	2,807	$2,\!358$	
1991	10,880	7,073	3,839	2,860	11,382	$6,\!352$	2,846	2,369	
1992	10,591	7,196	3,790	2,884	11,403	$6,\!541$	2,813	2,319	
1993	11,091	7,871	4,307	$3,\!106$	$13,\!621$	8,420	3,008	2,542	
1994	12,015	8,511	4,623	3,383	12,540	8,285	3,262	$2,\!657$	
1995	10,768	7,815	4,295	$3,\!240$	11,118	$7,\!582$	3,201	$2,\!537$	
1996	10,805	8,028	4,361	3,377	11,680	$7,\!582$	3,074	2,526	
1997	10,984	8,477	4,519	$3,\!511$	10,368	8,094	$3,\!114$	$2,\!544$	
1998	13,134	8,919	4,863	$3,\!446$	12,938	7,905	$3,\!176$	$2,\!494$	
1999	13,259	9,265	4,809	3,386	$10,\!640$	$7,\!441$	$3,\!154$	2,471	
2000	13,027	9,284	4,842	3,506	11,403	8,264	3,053	2,298	
2001	11,974	9,360	5,242	$3,\!436$	9,888	7,031	2,762	$2,\!114$	
2002	11,229	9,207	5,035	$3,\!488$	10,172	6,848	2,902	$2,\!111$	
2003	$11,\!669$	8,761	5,047	$3,\!487$	9,600	6,344	$2,\!657$	2,054	
2004	11,464	8,382	4,909	$3,\!435$	9,486	6,999	2,766	$2,\!123$	
2005	11.627	8,476	5.076	3.627	9.231	6.843	2,770	2.226	

#### Appendix 3.8 (continued) Average wage per-hour by occupational category (2005 Pesos)

Appendix $3.9$ (continued)							
Average wage per-hour in the public sector by occupational category							
(2005  Pesos)							

Year		Male				Female		
	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	12,829	9,111	4,706	$3,\!270$	10,316	6,430	4,227	2,818
1985	$11,\!497$	8,184	$4,\!373$	$3,\!288$	9,596	6,049	$3,\!914$	2,793
1986	$10,\!623$	8,292	$4,\!591$	$3,\!282$	8,399	6,023	$3,\!978$	$2,\!953$
1987	$12,\!910$	8,199	4,207	$3,\!152$	8,204	$6,\!284$	4,026	2,747
1988	$12,\!271$	$8,\!437$	$4,\!385$	$3,\!060$	8,533	$6,\!105$	$3,\!816$	$2,\!602$
1989	$11,\!365$	8,211	4,267	$3,\!117$	9,118	$6,\!278$	3,797	$2,\!547$
1990	10,808	$7,\!822$	$3,\!958$	3,022	8,718	6,093	3,730	$2,\!474$
1991	11,087	8,177	$3,\!955$	$3,\!065$	10,228	6,065	3,734	$2,\!392$
1992	$10,\!537$	7,732	$3,\!905$	$3,\!055$	$10,\!699$	6,739	$3,\!699$	$2,\!485$
1993	$11,\!674$	8,942	$4,\!427$	3,329	9,218	7,003	4,204	$2,\!579$
1994	$11,\!381$	10,297	$5,\!323$	$3,\!499$	$13,\!258$	7,029	4,060	$3,\!084$
1995	$10,\!973$	8,996	$4,\!591$	$3,\!332$	$10,\!325$	$6,\!898$	4,090	$3,\!011$
1996	$11,\!281$	9,063	4,531	$3,\!470$	9,560	$7,\!190$	4,241	$3,\!151$
1997	11,784	9,358	4,948	$3,\!635$	9,440	$7,\!816$	4,208	$3,\!246$
1998	$15,\!071$	$10,\!152$	4,920	$3,\!472$	9,285	$7,\!962$	4,826	$3,\!388$
1999	$15,\!338$	$10,\!380$	$5,\!278$	$3,\!423$	9,589	8,366	$4,\!472$	$3,\!300$
2000	$13,\!978$	10,278	$5,\!197$	$3,\!537$	10,737	$8,\!574$	$4,\!623$	$3,\!435$
2001	$12,\!678$	$10,\!059$	$5,\!486$	$3,\!416$	$10,\!850$	$8,\!864$	$5,\!090$	$3,\!485$
2002	10,272	10,122	$5,\!171$	$3,\!479$	$12,\!522$	$8,\!553$	4,952	$3,\!510$
2003	$11,\!843$	9,708	$5,\!489$	$3,\!493$	$11,\!310$	8,076	4,749	$3,\!467$
2004	12,760	9,385	$5,\!291$	$3,\!422$	9,379	7,721	$4,\!662$	$3,\!470$
2005	$12,\!396$	9,248	5,312	$3,\!689$	10,211	$7,\!958$	4,904	3,423

Appendix 3.10 (continued)
Average wage per-hour in the private sector by occupational category
(2005  Pesos)

Year		Male				Female		
rour	Managerial	Professional	Office	Other	Managerial	Professional	Office	Other
1984	14,114	8,822	3,419	2,685	11,558	5,990	3,308	2,291
1985	12,429	8,280	3,128	2,559	7,616	5,517	3,070	2,222
1986	11,938	8,723	3.172	2,625	8.075	5.141	3.072	2,256
1987	12,793	8,079	3,028	2,489	9,701	5,356	2,991	2,256
1988	12,971	8,085	2,826	2,430	8,338	5,270	2,870	2,131
1989	$12,\!134$	7,771	2,855	2,424	9,159	$5,\!595$	2,801	2,214
1990	12,109	7,378	2,806	$2,\!453$	8,560	5,105	2,808	$2,\!136$
1991	$12,\!341$	7,390	2,774	2,405	9,373	5,202	2,900	2,293
1992	12,230	7,460	2,820	2,400	9,750	5,512	2,808	$2,\!149$
1993	15,261	10,733	2,933	$2,\!649$	11,042	6,028	3,063	2,326
1994	13,725	9,510	3,213	2,765	$10,\!672$	6,965	3,299	$2,\!433$
1995	12,088	8,995	3,278	$2,\!649$	9,771	6,071	3,146	2,315
1996	$13,\!545$	8,908	3,123	$2,\!602$	9,102	6,210	3,038	$2,\!376$
1997	11,474	9,519	3,074	$2,\!676$	8,946	$6,\!579$	$3,\!145$	2,282
1998	14,477	9,223	3,262	2,583	10,870	$6,\!595$	$3,\!114$	2,322
1999	$11,\!815$	$8,\!425$	$3,\!170$	2,540	9,307	$6,\!539$	3,142	$2,\!339$
2000	$12,\!192$	$10,\!171$	$3,\!082$	$2,\!334$	10,361	6,701	3,033	2,234
2001	$11,\!148$	7,755	$2,\!644$	$2,\!151$	8,301	6,367	$2,\!847$	2,049
2002	$11,\!597$	$7,\!340$	$2,\!635$	$2,\!138$	8,532	6,416	$3,\!097$	2,064
2003	$10,\!887$	7,217	2,510	2,070	8,108	$5,\!608$	2,768	2,027
2004	$10,\!603$	8,080	$2,\!614$	$2,\!155$	8,040	6,084	$2,\!879$	2,070
2005	10,233	7,711	$2,\!617$	2,249	$8,\!130$	6,041	2,883	$2,\!189$

Year			Mana	gerial			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	5,902	15,214	14,115	11,901	$15,\!382$	8,712	10,757
1985	9,805	$9,\!186$	14,068	9,816	$9,\!370$	6,464	$7,\!421$
1986	9,532	$15,\!445$	$11,\!299$	10,236	$^{8,824}$	$8,\!485$	$8,\!665$
1987	9,080	11,099	$14,\!354$	$12,\!315$	$11,\!351$	9,192	8,288
1988	9,875	$14,\!812$	$11,\!510$	16,724	10,083	$10,\!273$	10,826
1989	$9,\!647$	$14,\!672$	12,202	$10,\!225$	$7,\!474$	$10,\!173$	9,724
1990	9,403	9,184	$11,\!993$	$11,\!155$	9,504	$9,\!411$	$7,\!394$
1991	$9,\!435$	11,903	12,917	14,212	$^{8,151}$	9,220	$9,\!449$
1992	$11,\!273$	7,745	$11,\!995$	$10,\!550$	$11,\!237$	9,165	$7,\!843$
1993	$7,\!832$	$6,\!891$	11,868	$11,\!637$	$25,\!170$	$7,\!585$	$^{8,520}$
1994	9,205	9,066	$18,\!375$	$9,\!433$	$10,\!213$	10,222	8,067
1995	$10,\!629$	15,787	$11,\!880$	$13,\!920$	$8,\!437$	$9,\!893$	9,203
1996	$11,\!116$	8,209	$11,\!441$	$9,\!276$	$11,\!061$	$13,\!236$	$9,\!805$
1997	10,362	$8,\!605$	14,767	10,704	10,913	12,790	9,569
1998	9,952	$18,\!086$	$14,\!610$	$14,\!947$	$12,\!466$	$23,\!144$	$11,\!250$
1999	$9,\!654$	$10,\!278$	$21,\!982$	$13,\!626$	$13,\!153$	11,783	$11,\!389$
2000	10,823	$39,\!915$	$15,\!558$	11,502	$11,\!128$	$27,\!427$	9,789
2001	10,252	$13,\!668$	$17,\!641$	$10,\!516$	$10,\!198$	$10,\!624$	$11,\!481$
2002	9,224	10,165	$17,\!466$	$14,\!205$	9,214	$10,\!690$	$10,\!956$
2003	$12,\!414$	9,327	17,746	10,034	11,518	10,404	$10,\!823$
2004	12,827	8,414	$19,\!105$	$13,\!119$	12,068	8,309	$11,\!357$
2005	9,702	$10,\!559$	$17,\!884$	11,502	$11,\!839$	9,582	10,922

# $\begin{array}{c} \mbox{Appendix 3.11 (continued)} \\ \mbox{Average wage per-hour in the public sector by occupational category} \\ \mbox{(2005 Pesos)} \end{array}$

Year			Mana	gerial			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	$\mathbf{Pasto}$
1984	9,361	20,427	14,328	14,870	12,597	10,182	7,803
1985	8,709	$6,\!935$	$12,\!329$	$11,\!003$	$11,\!315$	$9,\!175$	9,272
1986	7,026	20,498	$11,\!973$	$11,\!186$	$10,\!188$	$7,\!830$	8,705
1987	10,085	11,702	$13,\!143$	$15,\!287$	$9,\!171$	$11,\!191$	6,735
1988	8,707	$15,\!850$	11,795	$13,\!199$	$12,\!229$	$13,\!456$	8,054
1989	$7,\!829$	$10,\!372$	$12,\!925$	$10,\!839$	$10,\!549$	12,787	$7,\!604$
1990	$10,\!434$	$11,\!147$	12,040	$10,\!195$	$10,\!666$	11,520	$6,\!146$
1991	9,259	$13,\!399$	$13,\!186$	$10,\!894$	9,793	$14,\!220$	5,268
1992	$7,\!878$	$13,\!010$	$12,\!857$	9,984	11,027	$9,\!668$	8,061
1993	6,941	$11,\!986$	$15,\!263$	$17,\!310$	$14,\!073$	9,929	$8,\!051$
1994	6,778	9,741	$18,\!155$	8,733	$11,\!611$	$11,\!991$	$7,\!607$
1995	6,028	10,209	$16,\!134$	$11,\!676$	8,755	$10,\!583$	$7,\!186$
1996	7,204	$4,\!426$	$13,\!542$	11,012	13,797	14,792	8,718
1997	7,868	$5,\!038$	$14,\!265$	$11,\!488$	8,727	$14,\!588$	6,773
1998	7,776	$6,\!147$	18,917	11,798	$11,\!249$	$13,\!921$	$11,\!219$
1999	$6,\!171$	8,029	$15,\!399$	$10,\!492$	$8,\!816$	$11,\!976$	$8,\!199$
2000	7,076	$10,\!644$	$15,\!375$	$11,\!241$	11,704	8,935	8,565
2001	$6,\!423$	8,904	$15,\!094$	9,927	8,754	$11,\!391$	$8,\!984$
2002	7,066	7,918	$16,\!436$	$10,\!137$	9,795	9,323	$8,\!154$
2003	$6,\!297$	$7,\!833$	$13,\!609$	9,323	10,401	$9,\!636$	8,084
2004	$6,\!370$	6,936	$13,\!808$	9,510	9,982	9,721	6,325
2005	6,916	7,080	$14,\!331$	10,015	8,702	$7,\!693$	6,086

# $\begin{array}{c} \mbox{Appendix 3.12 (continued)} \\ \mbox{Average wage per-hour in the private sector by occupational category} \\ \mbox{(2005 Pesos)} \end{array}$

Year			Profes	sional			
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto
1984	7,439	7,929	8,077	8,383	7,202	7,157	7,307
1985	6,362	$6,\!673$	7,705	$7,\!483$	$6,\!820$	$6,\!395$	6,730
1986	$6,\!315$	$7,\!353$	$7,\!815$	$7,\!465$	6,724	$6,\!243$	$6,\!696$
1987	6,400	6,754	$7,\!846$	$7,\!691$	6,778	$6,\!584$	$7,\!140$
1988	$6,\!545$	$6,\!125$	7,929	$7,\!545$	$7,\!104$	$6,\!624$	$6,\!980$
1989	$6,\!561$	$6,\!889$	$8,\!073$	$7,\!514$	$6,\!695$	$6,\!988$	$6,\!482$
1990	$6,\!413$	$6,\!549$	7,766	$7,\!371$	6,809	$6,\!272$	$6,\!525$
1991	$6,\!377$	$6,\!400$	8,528	$7,\!093$	$7,\!112$	6,102	$6,\!199$
1992	6,333	9,740	$7,\!601$	$7,\!333$	7,074	$6,\!284$	$5,\!904$
1993	$6,\!925$	$7,\!195$	8,708	$11,\!432$	7,522	$7,\!371$	6,795
1994	$7,\!337$	8,013	$9,\!492$	9,044	9,248	$7,\!483$	$6,\!994$
1995	7,996	$7,\!544$	$8,\!949$	$^{8,267}$	6,967	$7,\!436$	$7,\!109$
1996	7,960	$7,\!544$	$9,\!495$	8,759	7,716	7,773	$7,\!361$
1997	8,562	$7,\!570$	$10,\!610$	$8,\!599$	$^{8,650}$	$8,\!561$	$7,\!603$
1998	9,730	$7,\!958$	10,703	9,525	9,263	$8,\!475$	$7,\!834$
1999	9,251	8,644	11,368	$9,\!843$	9,063	9,003	$^{8,507}$
2000	8,758	9,426	$10,\!035$	9,783	9,517	$9,\!482$	8,568
2001	9,720	$8,\!665$	$10,\!853$	9,389	10,220	$9,\!655$	$8,\!124$
2002	9,804	$8,\!478$	$10,\!470$	9,286	$9,\!612$	$8,\!978$	$8,\!668$
2003	8,990	8,206	$9,\!884$	9,343	9,301	8,010	$^{8,427}$
2004	8,705	7,003	9,840	$9,\!499$	$8,\!425$	8,703	$7,\!406$
2005	9,135	6,572	10,042	9,169	8,363	8,542	8,009

# $\begin{array}{c} \mbox{Appendix 3.13 (continued)} \\ \mbox{Average wage per-hour in the public sector by occupational category} \\ \mbox{(2005 Pesos)} \end{array}$

Year			Profess	sional			
	Bucaramanga	Barranquilla	Bogotá	$\operatorname{Cali}$	Medellín	Manizales	Pasto
1984	5,505	7,037	8,071	$7,\!586$	7,729	7,121	6,601
1985	5,852	$6,\!155$	7,588	$7,\!305$	$7,\!425$	$7,\!133$	$5,\!442$
1986	$5,\!448$	6,820	$7,\!548$	$6,\!988$	$7,\!898$	6,462	$4,\!491$
1987	$5,\!316$	$5,\!579$	$7,\!437$	$7,\!446$	$6,\!359$	$6,\!877$	$6,\!104$
1988	$5,\!143$	4,987	7,775	$7,\!074$	$6,\!514$	6,288	$4,\!474$
1989	$5,\!355$	5,362	$7,\!684$	$6,\!950$	$6,\!602$	$5,\!574$	$5,\!025$
1990	$5,\!489$	4,737	7,422	6,511	$6,\!547$	5,749	$4,\!681$
1991	$5,\!199$	4,747	7,711	$6,\!462$	$6,\!236$	$5,\!810$	$3,\!997$
1992	$5,\!377$	$5,\!143$	$7,\!646$	6,368	6,708	6,316	$4,\!447$
1993	5,780	6,101	11,775	7,717	$7,\!211$	$7,\!124$	$4,\!449$
1994	$6,\!392$	7,365	10,266	$7,\!256$	$7,\!596$	$6,\!666$	$5,\!016$
1995	$6,\!172$	5,721	9,028	$8,\!350$	$7,\!252$	6,755	$4,\!665$
1996	$6,\!668$	$5,\!849$	9,029	7,004	$8,\!147$	7,742	$6,\!438$
1997	6,921	6,211	10,756	$8,\!483$	8,640	7,218	$5,\!659$
1998	7,263	$6,\!454$	10,286	$7,\!650$	$7,\!808$	$7,\!477$	$5,\!950$
1999	$6,\!231$	6,290	8,977	$^{8,156}$	$7,\!263$	$7,\!948$	$5,\!548$
2000	8,749	6,889	$11,\!136$	$6,\!656$	$8,\!590$	$6,\!583$	$6,\!887$
2001	6,562	$6,\!480$	$8,\!639$	$6,\!546$	$7,\!673$	$6,\!552$	5,566
2002	$5,\!435$	$5,\!646$	$9,\!359$	$7,\!320$	6,773	6,010	$5,\!870$
2003	5,500	5,109	7,751	6,506	$7,\!550$	$5,\!423$	$5,\!450$
2004	5,974	5,007	$9,\!453$	$7,\!285$	$7,\!673$	$5,\!673$	$5,\!583$
2005	6,171	5,406	8,367	7,304	7,327	6,016	5,861

## Appendix 3.14 (continued) Average wage per-hour in the private sector by occupational category (2005 Pesos)

Year	Office							
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto	
1984	4,054	4,606	4,719	$4,\!190$	$4,\!153$	4,599	$3,\!804$	
1985	3,858	4,508	4,337	$3,\!901$	$3,\!913$	4,260	$3,\!614$	
1986	$4,\!103$	4,284	4,526	4,274	$4,\!189$	$3,\!811$	$3,\!934$	
1987	$3,\!817$	3,704	4,440	$4,\!453$	$3,\!996$	$3,\!804$	$3,\!595$	
1988	4,006	$3,\!819$	4,489	4,294	$3,\!814$	$3,\!649$	$3,\!487$	
1989	3,749	3,852	4,265	4,347	4,014	$3,\!977$	$3,\!370$	
1990	$3,\!875$	$3,\!844$	4,044	3,864	$3,\!820$	$3,\!641$	$3,\!466$	
1991	3,566	3,863	4,269	$3,\!954$	3,740	$3,\!426$	$3,\!378$	
1992	$3,\!424$	$3,\!681$	4,091	4,003	4,073	$3,\!658$	$3,\!180$	
1993	4,124	4,307	4,376	4,547	4,990	$3,\!976$	$3,\!469$	
1994	$4,\!172$	4,483	5,712	4,335	4,410	$3,\!814$	$3,\!422$	
1995	4,089	4,424	4,966	4,533	4,024	4,013	$3,\!193$	
1996	4,631	$5,\!298$	4,448	$3,\!997$	4,510	$3,\!815$	$3,\!810$	
1997	$4,\!372$	5,233	5,111	4,402	4,745	4,081	$3,\!933$	
1998	5,015	5,884	4,554	$4,\!615$	4,830	4,797	$4,\!433$	
1999	4,169	5,747	5,056	4,738	4,789	5,046	4,085	
2000	4,794	$5,\!637$	4,505	4,485	4,852	$5,\!425$	4,131	
2001	4,994	$5,\!642$	$5,\!196$	$5,\!428$	5,231	5,749	$4,\!383$	
2002	4,281	4,802	$6,\!656$	4,792	4,686	4,980	4,502	
2003	5,310	4,413	5,819	4,493	4,839	$5,\!196$	4,574	
2004	5,024	4,830	5,038	$5,\!199$	4,806	4,726	5,003	
2005	4,819	4,431	6,307	5,535	4,826	4,555	4,978	

# $\begin{array}{c} \mbox{Appendix 3.15 (continued)} \\ \mbox{Average wage per-hour in the public sector by occupational category} \\ \mbox{(2005 Pesos)} \end{array}$

Year	Office							
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto	
1984	2,898	3,555	$3,\!582$	$3,\!275$	$3,\!101$	$2,\!979$	$2,\!393$	
1985	$2,\!622$	$3,\!153$	$3,\!402$	$2,\!905$	$3,\!057$	$2,\!638$	2,224	
1986	2,704	3,087	3,463	3,022	$2,\!958$	$2,\!877$	$2,\!373$	
1987	$2,\!541$	2,776	$3,\!297$	$3,\!184$	2,900	2,731	$2,\!445$	
1988	$2,\!499$	2,588	3,083	2,977	2,756	$2,\!959$	$2,\!254$	
1989	$2,\!397$	2,764	3,099	2,913	2,722	$2,\!875$	2,111	
1990	$2,\!430$	2,703	$3,\!128$	2,743	$2,\!801$	2,586	$2,\!117$	
1991	$2,\!487$	2,743	3,269	$2,\!845$	$2,\!699$	2,559	2,074	
1992	2,417	2,708	3,166	2,787	2,753	2,559	2,238	
1993	$2,\!613$	2,842	3,368	$3,\!183$	2,935	2,814	2,246	
1994	2,901	3,292	3,702	$3,\!195$	3,001	2,878	2,260	
1995	$2,\!481$	3,013	3,933	$3,\!050$	2,984	$2,\!639$	$2,\!191$	
1996	$2,\!648$	3,528	$3,\!495$	2,913	3,006	2,732	$2,\!290$	
1997	2,743	3,359	3,844	2,999	$3,\!179$	2,765	2,339	
1998	2,840	$3,\!682$	$3,\!650$	2,974	3,161	2,877	2,335	
1999	$2,\!637$	3,377	3,827	$3,\!119$	$3,\!137$	2,952	$2,\!457$	
2000	$2,\!653$	3,734	3,425	2,760	2,982	2,973	$2,\!350$	
2001	2,502	2,947	3,079	2,739	2,734	2,802	2,264	
2002	2,495	2,849	3,914	2,896	2,731	2,585	2,326	
2003	2,406	2,437	$3,\!181$	$2,\!656$	2,759	$2,\!628$	2,240	
2004	2,454	2,591	3,349	2,824	2,869	2,483	2,304	
2005	2,496	2.673	3.386	2.795	2,758	2,491	2.207	

## Appendix 3.16 (continued) Average wage per-hour in the private sector by occupational category (2005 Pesos)

Year	Other							
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto	
1984	3,400	3,415	3,116	3,243	$3,\!173$	2,825	2,837	
1985	$3,\!344$	3,736	3,063	$3,\!279$	$3,\!276$	2,517	$2,\!657$	
1986	$3,\!151$	$3,\!118$	3,416	$3,\!103$	$3,\!379$	$2,\!629$	2,788	
1987	2,994	$3,\!378$	3,065	$3,\!238$	3,010	2,737	$2,\!656$	
1988	$3,\!134$	3,067	2,936	$3,\!078$	2,982	$2,\!458$	$2,\!675$	
1989	2,858	3,027	2,942	3,024	$3,\!097$	2,825	$2,\!698$	
1990	$2,\!893$	3,017	2,822	2,945	$2,\!936$	2,525	$2,\!665$	
1991	2,928	3,010	2,833	$2,\!831$	$3,\!083$	$2,\!499$	2,523	
1992	$3,\!058$	2,920	2,904	$3,\!072$	2,820	2,534	2,743	
1993	$3,\!237$	$3,\!274$	$3,\!087$	$3,\!405$	2,921	$2,\!672$	$3,\!083$	
1994	3,529	$3,\!470$	$3,\!428$	3,528	$3,\!432$	$3,\!147$	2,885	
1995	$3,\!477$	$3,\!317$	3,314	$3,\!492$	$3,\!135$	2,746	3,019	
1996	3,758	$3,\!627$	$3,\!420$	$3,\!097$	$3,\!434$	$3,\!128$	3,048	
1997	$3,\!532$	4,095	4,187	$3,\!405$	$3,\!429$	$3,\!166$	3,019	
1998	$3,\!641$	3,866	$3,\!478$	$3,\!291$	3,505	3,086	$3,\!357$	
1999	$3,\!458$	3,374	$3,\!685$	3,230	$3,\!494$	$3,\!156$	3,383	
2000	$4,\!147$	$3,\!418$	3,344	3,327	$3,\!801$	$3,\!410$	3,222	
2001	$3,\!374$	3,348	$3,\!233$	$3,\!492$	$3,\!561$	$3,\!668$	$3,\!274$	
2002	$3,\!562$	3,290	$3,\!370$	$3,\!905$	$3,\!668$	$3,\!287$	$3,\!492$	
2003	$3,\!606$	3,222	$3,\!400$	3,735	$3,\!618$	$3,\!367$	$3,\!527$	
2004	$3,\!831$	3,323	$3,\!067$	4,048	$3,\!541$	$3,\!112$	$3,\!525$	
2005	3,785	3,202	3,791	3,858	3,961	3,292	3,345	

## Appendix 3.17 (continued) Average wage per-hour in the public sector by occupational category (2005 Pesos)

Year	Other							
	Bucaramanga	Barranquilla	Bogotá	Cali	Medellín	Manizales	Pasto	
1984	2,454	2,705	$2,\!645$	2,667	2,589	$2,\!350$	1,648	
1985	$2,\!132$	2,722	2,566	$2,\!442$	$2,\!602$	1,965	$1,\!671$	
1986	2,269	2,548	2,779	$2,\!588$	2,501	$1,\!994$	$1,\!626$	
1987	2,209	2,264	2,574	$2,\!685$	$2,\!434$	2,037	$1,\!664$	
1988	2,120	2,225	2,477	$2,\!476$	$2,\!406$	$2,\!131$	$1,\!594$	
1989	2,214	2,286	2,539	2,523	2,363	$2,\!055$	$1,\!559$	
1990	2,168	2,311	$2,\!650$	$2,\!461$	$2,\!333$	1,974	1,516	
1991	2,168	$2,\!257$	2,715	$2,\!607$	$2,\!285$	1,967	1,513	
1992	2,160	2,250	2,522	$2,\!602$	$2,\!308$	2,007	$1,\!490$	
1993	2,270	2,369	2,791	$2,\!938$	$2,\!545$	2,004	$1,\!628$	
1994	$2,\!399$	2,733	2,978	$2,\!857$	2,506	$2,\!196$	$1,\!663$	
1995	$2,\!325$	2,556	2,899	2,734	$2,\!432$	1,937	$1,\!598$	
1996	2,360	2,588	2,930	2,501	$2,\!573$	$2,\!195$	1,713	
1997	2,322	$2,\!614$	$3,\!520$	$2,\!455$	$2,\!638$	$2,\!191$	1,752	
1998	2,319	2,562	$3,\!078$	$2,\!583$	2,562	$2,\!297$	1,757	
1999	2,243	2,593	2,906	$2,\!600$	$2,\!457$	$2,\!375$	1,866	
2000	2,054	2,386	$2,\!678$	$2,\!342$	$2,\!449$	2,289	$1,\!669$	
2001	1,952	2,308	2,286	2,313	2,281	2,016	$1,\!542$	
2002	2,038	2,307	2,340	$2,\!276$	$2,\!186$	1,907	$1,\!688$	
2003	1,923	2,165	2,185	$2,\!193$	$2,\!199$	$1,\!993$	$1,\!618$	
2004	1,981	2,261	2,290	2,201	2,259	2,074	$1,\!639$	
2005	2.077	2.362	2.465	2.368	2.330	2.034	1.706	

## Appendix 3.18 (continued) Average wage per-hour in the private sector by occupational category (2005 Pesos)